

RIVM report 481505020

#### Technical Report on Methodology: Cost Benefit Analysis and Policy Responses

D.W. Pearce, A. Howarth

May 2000

This Report has been prepared by RIVM, EFTEC, NTUA and IIASA in association with TME and TNO under contract with the Environment Directorate-General of the European Commission.

RIVM, P.O. Box 1, 3720 BA Bilthoven, telephone: 31 - 30 - 274 91 11; telefax: 31 - 30 - 274 29 71

## Abstract

The economic assessment of priorities for a European environmental policy plan focuses on twelve identified Prominent European Environmental Problems such as climate change, chemical risks and biodiversity. The study, commissioned by the European Commission (DG Environment) to a European consortium led by RIVM, provides a basis for priority setting for European environmental policy planning in support of the sixth Environmental Action Programme as follow-up of the current fifth Environmental Action Plan called 'Towards Sustainability'. The analysis is based on an examination of the cost of avoided damage, environmental expenditures, risk assessment, public opinion, social incidence and sustainability. The study incorporates information on targets, scenario results, and policy options and measures including their costs and benefits.

Main findings of the study are the following. Current trends show that if all existing policies are fully implemented and enforced, the European Union will be successful in reducing pressures on the environment. However, damage to human health and ecosystems can be substantially reduced with accelerated policies. The implementation costs of these additional policies will not exceed the environmental benefits and the impact on the economy is manageable. This requires future policies to focus on least-cost solutions and follow an integrated approach. Nevertheless, these policies will not be adequate for achieving all policy objectives. Remaining major problems are the excess load of nitrogen in the ecosystem, exceedance of air quality guidelines (especially particulate matter), noise nuisance and biodiversity loss.

This report is one of a series supporting the main report: *European Environmental Priorities: an Integrated Economic and Environmental Assessment*. The areas discussed in the main report are fully documented in the various *Technical reports*. A background report is presented for each environmental issue giving an outline of the problem and its relationship to economic sectors and other issues; the benefits and the cost-benefit analysis; and the policy responses. Additional reports outline the benefits methodology, the EU enlargement issue and the macro-economic consequences of the scenarios. Technical Report on Methodology: Cost Benefit Analysis and Policy Responses

This report has been prepared by RIVM, EFTEC, NTUA and IIASA in association with TME and TNO under contract with the Environment Directorate-General of the European Commission. This report is one of a series supporting the main report titled *European Environmental Priorities: an Integrated Economic and Environmental Assessment.* Reports in this series have been subject to limited peer review.

Prepared by D.W. Pearce, A. Howarth (EFTEC)

The following sections are the supporting documents to the benefit assessment and policy assessment papers in the main report. Section 1 describes the benefit assessment procedure applied to the eleven environmental issues in this study. Section 2 concentrates on the nature of economic instruments and the criteria used to select economic instruments. This is followed by a brief typology of economic instruments and finally this section makes the first step towards matching policies to the environmental issues considered in this study. Section 3 introduces monetary valuation of 'non-marketed' environmental goods, such as clean air, clean water, etc. The concept of 'total economic value' is discussed, followed by a brief description of the valuation techniques used. This study relies heavily on the process of 'benefits transfer' (BT) which involves taking existing monetary valuation studies (i.e. 'willingness-to-pay' values) and applying them outside the site context where the study was originally conducted. Section 4 describes the adjustments involved in the benefits transfer process and lists the main criteria for successful and accurate benefits transfer. Section 5 presents the analytics and the empirical evidence of the income elasticity of demand for the environment. This information is used to calculate benefits in the future (i.e. in 2010), it is assumed that environmental quality has a rising relative price through time that is linked to growth in income per capita. Section 6 introduces the importance of risk valuation in environmental cost-benefit analysis. The concept of a 'value of statistical life' (VOSL) is discussed, a brief discussion of the techniques used to estimate VOSL is given followed by empirical evidence of the VOSL. This section also discusses the 'value of life year' (VOLY) as well as looking at others' valuation of risk to individuals, the value of future lives and the affect of unequal income distribution on the VOSL.

The findings, conclusions, recommendations and views expressed in this report represent those of the authors and do not necessarily coincide with those of the European Commission services.

## Contents

1. BENEFIT ASSESSMENT PROCEDURE	7
2. POLICIES AND ENVIRONMENTAL PROBLEMS	13
3. MONETARY VALUATION TECHNIQUES	23
4. BENEFITS TRANSFER	31
5. INCOME ELASTICITY OF DEMAND FOR THE ENVIRONMENT	35
6. VALUING STATISTICAL LIVES	37
References	56

#### **1. BENEFIT ASSESSMENT PROCEDURE**

#### **General methodology**

The general structure of the benefit assessments for the different environmental problems is as follows:

#### Importance of the issue

A brief discussion stating how public and expert opinion rank the issue as a serious environmental problem is provided.

#### Monetary valuation

The measurement of benefits is essentially the measurement of avoided damages. Since the scenarios (other than Accession) generally simulate overall improvements in the environment, benefits will tend to get larger as we move from the baseline to the AP or TD scenarios. Where D refers to environmental damage, we estimate the benefits for the AP and TD scenarios in the following way:

 $D_{BASE} - D_{AP} = Benefits of AP$  $D_{BASE} - D_{TD} = Benefits of TD$ 

Figure 1.1 gives a stylised illustration of the benefit in 2010 of the AP scenario over baseline.



Figure 1.1: Benefit of AP scenario in 2010

Benefits are defined in terms of individuals' willingness to pay (WTP) to secure the benefits. Ideally a wider concept of benefits would include the macroeconomic benefits. Essentially, the difference between the WTP and macroeconomic benefits is one of scope. WTP tends to capture 'partial' benefits. These will approximate total benefits if the policy measures make only marginal changes to the economy. However, in terms of the scenarios adopted, the policy measures are not marginal, but involve fairly significant discrete changes. Thus, the wider concept of benefit is embraced allowing for the feedback effects of the policy on other prices and quantities in the economy.

This is possible only for those measures relating to the energy sector where the GEM-E3 model is used. In other cases, no macroeconomic model exists or the measures in question cannot easily be incorporated in such models (e.g. coastal waters, biodiversity etc).

Relevant WTP values for each environmental problems are drawn from an extensive literature review of the most recent and relevant monetary valuation studies conducted for this study. These give WTP estimates for environmental improvement or WTP estimates to avoid environmental damage. Two main groups of monetary valuation techniques are used, stated and revealed preference techniques (for further details refer to Section 3 on monetary valuation techniques). Taking existing monetary valuation (WTP) studies and applying them outside of the site contexts where the study was originally carried out requires extensive use of 'benefits transfer' (see Section 4 on benefits transfer).

In order to calculate benefits in the future (i.e. in 2010), we assume that environmental quality has a rising relative price through time that is linked to growth in income per capita. The following formula is adopted to adjust the valuations accordingly:

$$WTP_{2010} = WTP_{1990} \cdot (Y_{2010}/Y_{1990})^{e}$$

Where *WTP* refers to willingness to pay valuations, *Y* is EU per capita GNP (assumed as:  $Y_{2010} = \notin 19830$  and  $Y_{1990} = \notin 14247$ , source RIVM) and *e* is the income elasticity of demand (assumed here as 0.3 refer to Section 5 on income elasticity of demand). An annual increase in relative prices for environmental quality of 0.5% per annum is arrived at<sup>1</sup>.

The likely time paths of benefits (and costs) are known in only a few cases, thus the benefit results are reported for a representative future year only, i.e. 2010. It is acknowledged that different time paths will produce potentially different results. When benefits are later compared with costs, the net benefits clearly depend on <u>what</u> is done, the <u>scale</u> of control measures and on the <u>time paths</u> of these measures. However, the research team is of the view that no major divergence of results will occur because of the choice of a representative year for benefits and costs.

Benefit estimates are summarised in Table 1.1 for those environmental problems with clearly defined AP / TD scenarios, such as climate change, acidification, tropospheric ozone, waste management, Human health, air quality and noise and nuclear risks. The values relate to benefits to the EU15 only unless otherwise stated. All values are benefits in 2010 only and are given in terms of  $\in$  (1997 prices). For a more detailed discussion of the benefit estimates, refer to the benefits assessment for each environmental problem given in the Technical Reports.

<sup>&</sup>lt;sup>1</sup> Ideally, it should be decomposed by country since income per capita varies by country. However, we suggest EU-wide approximations are suitable.

Environmental problem	Primary benefit	Secondary benefit	
	€ billion	€ billion	
Climate change			
NT-AP	3.7*	20.5 (11.5)	
FT-AP	3.7*	13.4 (7.5)	
Acidification			
NT-AP	21.7 (14.0)	7.1 (1.6)	
FT-AP	25.2 (16.3)	7.8 (1.7)	
TD	58.9 (38.1)	12.6 (1.7)	
Tropospheric ozone			
NT-AP	5.6 (0.7)	-	
FT-AP	5.7 (0.7)	-	
TD	9.1 (1.2)	-	
Waste management			
AP with source reduction	8.7	0.4	
AP without source reduction	7.2	0.4	
TD max compost and recycle	10.3	-	
TD max incineration	-2.8	-	
Particulate matter			
AP	5.3 (3.1)	-	
TD	(24.2 (14.0))	-	
Nuclear risks			
TD	6.8	-	

 Table 1.1
 Summary of benefit estimates for each environmental problem

Where: '-' = not available, NT = No Trade, FT = Full Trade, AP = Accelerated Policy scenario, TD = technology driven scenario. \* benefit to world. Estimates given in brackets assume premature mortality is valued with VOLY, estimates not bracketed assume VOSL.

Note, primary benefits relate to the control of the pollutants,  $P_X$ , causing the environmental problem X. Primary benefits for climate change are due to the control of CO<sub>2</sub>, CH<sub>4</sub> and N<sub>2</sub>O only. For acidification, primary benefits are due to control of SO<sub>x</sub>, NO<sub>x</sub> and NH<sub>3</sub>, for tropospheric ozone primary benefit estimates are due to the direct control of VOCs only and for Human health, air quality and noise, primary benefits relate to the end-of-pipe measures that reduce concentrations of primary PM<sub>10</sub>.

However, the control of  $P_X$  pollutants can lead to the control of other pollutants Py causing other environmental problems Y. This effect is known as the secondary benefit of measures to control X. The secondary benefits of climate change control are to acidification, low level ozone and Human health, air quality and noise (through the control of primary PM<sub>10</sub> and secondary aerosols). Note that the secondary benefits due to reductions of secondary aerosols are not estimated separately, they are already accounted for in the secondary benefits to acidification. The secondary benefits of acidification control are to tropospheric ozone and Human health, air quality and noise (i.e. through the control of primary PM<sub>10</sub> and secondary aerosols). Note that the secondary benefits due to reductions in secondary aerosols are subsumed in the primary benefit estimates for acidification. The secondary benefits of waste management are to climate change control.

#### Uncertainty

Uncertainty is endemic to this study. The main sources of uncertainty in the benefit assessment approach are:

- scientific uncertainty about the impacts of given pressures on the state of the environment;
- economic uncertainty about the willingness to pay of the relevant population to avoid the impact;

Since these uncertainties are unavoidable the relevant approach is one which tries to estimate central tendencies and the confidence interval around that central tendency. Even though uncertainties at the various stages of the analysis may be multiplicative, estimates of central

tendency will tend to remain unchanged, although the dispersion about the mean will increase.

It is tempting to think that avoiding some of the stages of the analysis may reduce uncertainty. For example, monetary valuation (WTP) of premature mortality may be subject to confidence ranges which increases the range of uncertainty attached to the impact measure (e.g. lives lost). Casual commentators suggest that the monetary valuation stage should therefore be avoided. This is a mistaken strategy. First, the addition of the monetary valuation stage does not change the mean outcome. Second, the wider confidence interval that may emerge does indeed 'increase' the uncertainty of the estimate of effect, but if the monetary valuation stage is avoided other forms of uncertainty are added to the picture. Pursuing the pollution-health example, the analysis may be presented in terms of costs and lives prematurely lost, or it may be presented in terms of costs and the monetised (economic) value of lives prematurely lost. The former avoids the monetary valuation estimate; the latter explicitly includes it. But while the latter adds to uncertainty in the sense of increasing the confidence interval, the former increases other forms of uncertainty. Using the former, i.e. 'lives lost prematurely' assumes, for example, that all lives are to be equally weighted regardless of the length of life expectancy lost, or the health state of those at risk etc. If this is not what is desired, then lives lost prematurely can be weighted by life expectancy and health state. But in so doing, the analyst superimposes a weighing on the indicator that has nothing to do with the perceptions or preferences of those at risk. In short, a new form of uncertainty is introduced, namely the uncertainty about the extent to which indicators are now responsive to individuals' wants and desires, the basic value axiom of welfare economics.

Avoiding the monetary valuation stage also creates other forms of uncertainty. Where the impacts are measured in non-monetary units and compared to costs, there is no guideline on whether a policy is worth undertaking. Monetisation provides the guideline that policies should at least past a test to the effect that benefits should exceed costs. Cost-effectiveness indicators have no such test since it is never possible to tell whether an incremental unit of effectiveness is worth the cost of securing it. (Note also, that selecting any target of effectiveness, e.g. an upper limit on cost per life saved, automatically implies a monetary benefit estimate).

Avoiding the monetary valuation stage may seem like a rational response to the uncertainty embedded in the benefit estimates, but such a response adds at least two other forms of uncertainty. Firstly, the 'democratic' uncertainty, this is the extent to which any outcome is now responsive to individuals' preferences, and secondly the 'decision-making' uncertainty, i.e. the extent to which rational trade-offs between costs and benefits can be made.

The reliability of the WTP values used in this study is tested, where possible, by using confidence intervals around the mean value. Table 1.1 reports mid values only, for ranges refer to the benefit estimates for each environmental problem given in the Technical Reports.

The monetary valuation of premature mortality is a key area of uncertainty in this analysis. Where the probability of death is the same for all age groups in the population we adopt the value of a statistical life relevant to the general population, i.e.  $\in$  3.31 million (from  $\in$  2.6m 1990 prices converted to 1997 prices using the deflator 1.274). In those areas where deaths are mainly confined to the older age groups in the population we use a reduced VOSL, i.e. 70% of  $\in$  3.31 million =  $\in$  2.32 million.

For some environmental problems, fatalities occur over a long period, i.e. 1990-2010-2050 (i.e. nuclear risks) and thus the VOSL relevant to 1990 will not be relevant to the whole period. Rather, we would expect VOSL to rise as incomes rise. The effect of income growth is captured by introducing a rising relative price of risk aversion of 0.5% per annum, although there is limited information on the increase over time in the relative 'price' of risk. For further

information regarding the issue of premature mortality valuation see Section 6 on valuing statistical live.

#### Sensitivity

A number of assumptions are made for each separate benefit assessment. Some may have a significant effect on the results, while others will make only a minor difference. For purposes of transparency the key assumptions are stated clearly throughout. In order to see the effect on the net results if these assumptions are changed we conduct a sensitivity analysis. Thus, changes in the key assumptions and the associated quantitative effects are also reported.

## 2. POLICIES AND ENVIRONMENTAL PROBLEMS

Each environmental problem considered in this study contains a set of quantitative or qualitative 'targets' and each target corresponds to a scenario. Targets might be set in terms of given reductions in emissions, areas of land conserved for biodiversity etc. In order to inform the design of environmental policy, we need some idea of what policy instruments are best suited to the achievement of the targets.

#### The nature of policy instruments

A critical goal of policy towards the environment is cost-effectiveness: the achievement of the policy goal at least cost. In welfare economics terms, 'least cost' means least loss of economic wellbeing<sup>2</sup>. A narrower goal would be to measure costs solely in terms of the costs borne by the regulated agent in complying with the policy.

There are general reasons for supposing that *economic instruments* are best suited to achieving the least cost goal. Definitions of economic instruments (EIs) are not easy to provide. All forms of regulation impose a cost on the regulated agent, so that the presence of a financial incentive is not peculiar to economic instruments. It is widely argued that EIs leave the regulated agent with more flexibility on *how* to respond to policy. Thus, traditional 'command and control' (CAC) regulation might be regarded as setting target (what to achieve) and mechanism (how to achieve it), whereas EIs leave the regulated agent with the choice of what to achieve and how to achieve it, provided the overall policy goal is met in the aggregate. Thus, an individual regulated agent can emit pollution up to any level provided it pays the necessary environmental tax or holds the necessary permit to emit. The choice of the mix of abatement measures and tax payments/permit holdings is up to the regulated agent. But policy will have set an aggregate goal, for example a total level of emissions, that must be met and permits will be issued equal to this aggregate goal, or an estimate will have been made of the emission reduction effect of taxes so as to achieve the goal.

There are general reasons for supposing that EIs are best suited to achieving the least cost goal. However, the presumption that EIs are more cost effective than CAC is not always the case. In general, quite specific conditions have to be present for EIs to perform better than CAC<sup>3</sup>. These factors need to be taken into account in deciding the 'match' of policy instruments to environmental problems.

#### Criteria for selecting policy instruments

Fundamental to this study is the use of 'welfare economics', it is therefore appropriate that the criteria for selecting 'desirable' policy instruments should be based on social cost benefit analysis. However, it is important to assess policy instruments against other considerations, such as distributional concerns (i.e. impacts to socio-economic class and region), macroeconomic issues (competition and employment effects), administrative feasibility<sup>4</sup> and subsidiarity (i.e. the 'optimal jurisdiction' issue, in other words, where policy is most effectively located). Subjecting policy instruments to many criteria for acceptability risks making almost all policy instruments fail. Similarly, we have no clear criteria (meta-criteria) for deciding which criteria are the most important. In order to identify rational policy instruments to meet AP scenario targets, we suggest that there are five groups of criteria for choosing a policy instrument, these are set out below:

 $<sup>^{2}</sup>$  Which, ideally, would be measured by the change in the sum of producers' and consumers' surpluses. In practice, this measure will be available in only some cases.

<sup>&</sup>lt;sup>3</sup> See C.Russell, P.Powell and W Vaughan, Rethinking advice on environmental policy instrument choice in developing countries, Paper to World Congress on Environmental Economics, Venice, June 1998.

<sup>&</sup>lt;sup>4</sup> Note that, 'political feasibility' is not explicitly considered, since the research team's concern is to define a potential menu of policies. The extent to which such policies are politically feasible is not for the research team to judge.

- causal
- efficiency
- equity
- macro-economic
- jurisdictional

The <u>causal</u> criterion answers the basic question: 'does the policy instrument address the underlying economic failure'? If policy does not address the real causes of environmental degradation, it will have a high risk of failure. It is important to note that real causes do not equate with 'pressures' in the DPSR paradigm, nor what is popularly understood by 'driving forces'. The underlying causes are i) market failures (i.e. not well defined property rights, missing markets and lack of information; ii) intervention failures, i.e. if legislation exists, but is not fully implemented by Member States, iv) growth of real income, and v) population change, i.e. natural growth, migration and social change. Overall policy measures are targeted at the first three underlying causes only.

The economic <u>efficiency</u> criterion includes: i) benefit cost ratios, ii) cost-effectiveness, iii) benefits, and iv) public opinion for each policy instrument. The least-cost action is embodied in the cost-benefit approach and in cost effectiveness. Public opinion is included in efficiency because public opinion indicates public preferences, which in turn underlie the notion of willingness to pay. WTP is the building block of the benefits assessment.

The <u>equity</u> or distributional criterion considers: i) intra-generational equity (impacts to current socio-economic class, economic sector and region) and ii) inter-generational equity (distributional impacts between generations).

The <u>macro-economic</u> criterion is mainly centred on the NTUA modelling of climate change policy. Policy instrument impacts considered are EU employment, GNP loss and competition effects. The macro-economic impacts are in the final report, see *Chapter 4 Section 4.1*, and for further details refer to *Technical report on Socio-economic trends, macro-economic impacts and cost interface*.

The <u>jurisdictional</u> criterion concentrates on the issue of subsidiarity, i.e. where is policy most effectively located, such as, EU, national or local level. There are three main criteria upon which the level of subsidiarity can be assessed: i) gains from co-operation, ii) gains from harmonisation and co-operation and iii) gains in sustainable implementation.

#### **Types of instruments**

The list of instruments is potentially very large. Here we categorise them according to OECD classifications and the discussion in Panayotou  $(1998)^5$ .

#### Command and control

- ambient based standards (e.g.  $\mu$ g of pollutant per m<sup>3</sup>);
- emission based standards (e.g. g of pollutant per km travelled in test conditions);
- product based standards;
- technology based standards (BAT, BATNEEC), and
- bans.

5

See OECD, Managing the Environment: the Role of Economic Instruments, OECD, Paris, 1994, and T.Panayotou, Economic Instruments for Environmental Management and Sustainable Development, Earthscan, 1998, forthcoming.

Standards may be based on a per pollutant basis or on Integrated Pollution Control (IPC) considerations such that impacts on different environmental media are considered.

#### Economic instruments

#### **Property rights**

Property rights should be secure (enforceable) and transferable for economic efficiency to be assured. They will need to be attenuated in some form (ie certain uses will be forbidden) if they give rise to excessive externalities. Property rights can be private, communal or public, with a presumption that private and communal rights are to be preferred.

#### **Fiscal instruments**

- emission taxes (e.g. SO<sub>x</sub> charge);
- effluent taxes (e.g. wastewater charge);
- input taxes (e.g. pesticides or fertiliser tax);
- (final) product taxes (e.g. packaging tax);
- export taxes/import taxes;
- differential taxation (e.g. leaded/unleaded gasoline);
- royalty (rent) taxation (e.g. forest taxation);
- land use taxes (taxes vary according to land use);
- accelerated depreciation (environmentally beneficial investments allowed to depreciate faster for tax offset purposes);
- subsidy removal (where subsidies harm the environment, e.g. CAP reform), and
- subsidies (where subsidies benefit the environment) (e.g. subsidies to renewable energy).

Pollution taxes are formally equivalent to pollution charges so that no distinction between the two is made here. But charges and taxes otherwise differ: charges are for the use of a service, whereas taxes tend to raise revenue. The equivalence of charges and taxes in the pollution case arises because the polluter is using a public service - the assimilative capacity of the environment. In administrative terms the more important distinction is that taxes always form part of the fiscal structure and have therefore to be agreed by and administered by the tax authorities. Charges can be outside the control of tax authorities (inland revenue and customs).

#### **Environmental charges**

- user charges (e.g. entry fees to protected areas, road pricing);
- betterment charges (charges on properties which benefit from public infrastructure or environmental improvement), and
- impact charges (the obverse of betterment, ie charges on properties for making the environment worse, usually levied when property or land use changes).

#### **Deposit - refund schemes and performance bonds**

Here the charge is made in advance of any damage occurring, and refunds are given when the product is safely disposed of or recycled or the environmental degradation is made good. Bonds act in the same way: the bond has to be purchased at the onset on economic activity (e.g. quarrying) and can only be redeemed when there is an indication that restoration has occurred.

- deposit-refund (tax-subsidy) schemes (e.g. on returnable bottles and cans);
- environmental performance bond (e.g. mining, quarrying, forest logging, waste arisings), and
- accident bonds (e.g. for oil spills).

#### Liability systems

Liability systems rest on the threat of legal action in the event of non-compliance. The charge is collected only in the event of damage occurring and liability systems thus have similarities with

bonds (above). However, bonds collect the charge early on and refund it later. Liability systems collect the charge only in the event of damage.

- legal liability (is 'strict' when libaility exists regardless of precautions taken, and is 'negligence' when actions taken to avoid damage are taken into account);
- non-compliance fines (charges at some penal rate for emissions above standards);
- joint and several liability (any one contributor to damage can be held responsible for all damage), and
- liability insurance (insurance market premia in the event of damage in a liability context).

#### **Financial incentives**

Financial incentives involve the creation of funds used for environmental improvement. Funds may come directly from government grants, from specific taxes or from external 'deals' such as a debt-for-nature swap, Global Environment Facility incremental cost financing etc. Financial incentives are especially important for the Economies in Transition.

#### Tradable quotas and offsets

Tradable quotas can relate to emissions (tradable emission permits) or resources (tradable resource quotas). Offsets relate to bargains between several parties such that an emission reduction obligation in one location is offset by reducing emissions in another location. The credits may not be traded (joint implementation) or they may be traded (tradable emission credits).

- joint implementation (mainly CO<sub>2</sub> but not exclusively);
- tradable emission permits (SO<sub>x</sub> in the USA, and limited use in Europe): auctioned / grandfathered;
- tradable water rights;
- tradable fishing quotas: auctioned / grandfathered, and
- tradable development rights (land is zoned, some of it for development and rights to that development then become tradable).

#### Voluntary agreements

Voluntary agreements involve an understanding, sometimes backed by legal requirements, between government and industry such that industry 'self regulates'. Self regulation involves setting agreed environmental targets, leaving industry to determine its own means of achieving those targets, such to some overall broad agreement on mechanisms.

#### Information

Two forms of information provision are considered:

- labelling (labelling of environmental performance, resource content etc)
- disclosure (publication of pollution profile of companies etc)

#### Matching policies to environmental problems

The following matrices 'match' environmental problems and economic instruments based on the five criteria indicated above. The allocation is necessarily judgmental but conforms with exercises elsewhere that have attempted to link instruments and problems in various different contexts. Environmental funds are excluded from the analysis because they can be created through the other instruments. However, externally financed funds are of importance to 'economies in transition. Property rights are also excluded as they are of less concern in Europe. Nonetheless, each environmental problem is prefaced with a remark about property rights. These matrices are the foundations for the policy packages / assessments for each environmental issue presented in this Annex.

#### Stratospheric ozone depletion

Property rights in the ozone layer were established by the Montreal Protocol and the subsequent amendments and agreements. Rights are held by all signatory countries. Policy measures relate to controls on domestic production and controls on imports due to the fact that imported ODSs contribute to domestic consumption totals which, are the subject of 'caps'. Imports of recycled ODSs do not count against domestic consumption. *Financial incentives* (not shown here) relate to the Multilateral Fund which finances phase-out in the developing countries. Note also that the Montreal Protocol makes extensive use of restrictions on international trade in CFCs.

Stratospheric ozone depletion

Issue: reduce emissions of ODSs	
Initiatives	
Fiscal incentives	CFC taxes exist in USA
	Import duty reductions for ODS
Charges	-
Deposit refund schemes and performance bonds	DRS for recycled ODSs
Liability	-
Tradable permits	Reduction in trading in USA
	Permit trading for import allowances
Voluntary agreements	VAs to restrict imports
Information	Labelling products

#### **Climate change**

Property rights established by FCCC, 1992 and Kyoto Protocol 1997/8. Financing for LDC emissions reduction takes place via incremental cost financing from the Global Environment Facility, and through joint implementation schemes. The Clean Development Mechanism introduced in the Kyoto Protocol could evolve into a North-South JI scheme. JI East-West is enabled under the Kyoto Protocol.

Climate	change
---------	--------

Issue reduce emissions of GHGs and sequestering carbon	
Initiatives	
Fiscal incentives	Carbon / energy taxes in place in several
	countries;
	Excise duties;
	Aviation tax;
	Methane tax;
Charges	-
Deposit refund schemes and performance bonds	-
Liability	-
Tradable permits	Joint implementation in place: over 200 deals
	Tradable efficiency permits for car manufacturers
Voluntary agreements	Carbon neutral pricing schemes
	Voluntary offset schemes
Information	Energy conservation campaign
	Emission disclosure

#### **Major accidents**

Major accidents occurring in a single Member State and not affecting other States need to be distinguished from accidents with potential transboundary effects. Nuclear, oil sill and chemical risks can easily be transboundary, suggesting that preventive and emergency response measures should be co-ordinated at EU level. To be realistic, such measures need to incorporate funds, akin to the Montreal Protocol Multilateral Fund, to finance risk reduction in the EITs. Such a

fund exists for nuclear accidents (EU, Canada and US financed) and there are emergency response communications co-ordinated across Europe.

Major accidents		
Issues reducing high risk nuclear reactors, reducing chance and impact of chemical disasters		
Initiatives		
Fiscal incentives	Tax on energy output	
	Tax on port calls	
	Output tax	
	All to fund emergency responses	
Charges	-	
Deposit refund schemes and performance bonds	Could introduce performance bonds in EU	
Liability	Negligence liability in EU	
Tradable permits	-	
Voluntary agreements	-	
Information	-	

#### **Biodiversity loss**

Unless privately owned or legally protected, most biodiversity is not the subject of property rights. Ownership of land by conservation groups or the state can contribute substantially to reducing biodiversity loss. Effectively, a market in biodiversity is created, although the medium is the land and property market itself. In other cases, market creation may be direct, e.g. by commercialising products from wild species, creating an incentive to conserve the species for profit. Since pollution is a cause of biodiversity loss it should be noted that all pollution reduction measures (see other environmental problems) will have an impact on biodiversity.

Issue: reducing biodiversity loss through reduced h	abitat loss
Initiatives	
Fiscal incentives	Agri-environmental schemes; environmentally sensitive areas, country side access schemes, country side stewardship schemes, Arable stewardship schemes, habitat schemes, moorland schemes, organic farming schemes, nitrate sensitive areas, Payments for set-aside; Outright land purchases; Tax allowances on money and land donations to conservation; Tax allowances on reforesting, soil and water conservation, and Easements and purchase of development rights;
Charges	Park entrance fees, user permits with earmaked revenues; Fines for damage to natural assets.
Deposit refund schemes and performance bonds	Land restoration with performance bonds
Liability	Liability for pollution damage
Tradable permits	Offset requirements, e.g. loss of wetland has to be offset by creation of new wetlands, i.e. mitigation banking; Tradable development rights; Tradable fishing quotas.
Voluntary agreements	Voluntary management agreements: Sweden, Austria, UK.
Information	Ecolabelling

Biodiversity loss

#### Acidification and eutrophcation

Property rights to transboundary pollution reduction have been established by the Convention on Long Range Transport of Air Pollution in Europe and by various EU legislation.

Acidification	and	outrophication
Actuification	ana	europhication

Issue: reducing emissions of $SO_x$ , $NO_x$ and $NH_3$	
Initiatives	
Fiscal incentives	S and N taxes
	NH <sub>3</sub> tax with mineral accounting, or livestock tax
Charges	-
Deposit refund schemes and performance bonds	-
Liability	-
Tradable permits	Possible tradable permits in SO <sub>x</sub> and NO <sub>x</sub>
Voluntary agreements	-
Information	-

#### **Chemical risks**

Chemical risks Issue: reducing risks from heavy metals, pesticides and POPs Initiatives Fiscal incentives Pesticide tax Battery charges Chemicals charges Charges Deposit refund schemes and performance bonds Application to hazardous products, e.g. batteries Liability Tradable permits Lead trading Voluntary agreements VAs possible Information Ecolabelling

#### Water management

Water management

Issue: improving water availability through management of supply and demand, and improving quality of ground water and surface water. Initiatives Fiscal incentives Pesticide tax Fertiliser tax Abstraction charges Charges Effluent charges Deposit refund schemes and performance bonds -Liability \_ Tradable permits Tradable water rights Tradable effluent rights Tradable quotas for pesticides and fertilisers Voluntary agreements -Information

Note: main requirement is to control for water demand through pricing of water at long run marginal cost.

#### Waste management

Waste	management
music	munugemeni

Issue: reducing waste at source, increase recyling and re-use, minimise landfill	
Initiatives	
Fiscal incentives	Recycling credits;
	Virgin materials tax;
	Landfill tax;
	Incinertion tax.
Charges	Collection charges
Deposit refund schemes and performance bonds	DRSs for returnable containers;
	DRSs for batteries.
Liability	-
Tradable permits	Tradable recycling quotas
Voluntary agreements	Producer responsibility agreements
Information	-

### **Tropospheric ozone**

Many of the policies suitable for acidification will also have a significant impact on the problem of tropospheric ozone.

Tropospheric ozone		
Issue: reduce $NO_x$ and VOC emissions (i.e. the precursors to low level ozone)		
Initiatives		
Fiscal incentives	N tax	
	VOC tax	
Charges	-	
Deposit refund schemes and performance bonds	-	
Liability	-	
Tradable permits	Tradable quotas in VOCs and NO <sub>x</sub>	
Voluntary agreements	-	
Information	Eco-labelling for solvents	

#### **Coastal zone management**

See also climate change, biodiversity loss and chemical risks.

Coastal zone management

Issue: reduce coastal erosion (see climate change), reduce habitats damage (see biodiversity loss), improve bathing water quality		
Initiatives		
Fiscal incentives for:		
Bathing water quality	Tax non compliance with Bathing Water	
	Directive	
Charges	Possible beach charges	
Deposit refund schemes and performance bonds	-	
Liability	Owner liability for failure to meet Bathing Water	
	Directive;	
	Owner liability and performance bonds against oil	
	spills.	
Tradable permits	Transferable development rights;	
	Tradable quotas for fishing	
Voluntary agreements	-	
Information	-	

#### Human health, air quality and noise

#### Human health, air quality and noise

Issue: reduce exposure to noise, reduce urban pollutants especially PM10, PM2.5.		
Initiatives		
Fiscal incentives	Air pollution taxes: see acidification Noise taxes for vehicles Airport landing charges varied with noise levels;	
Charges	Road user charges according to noise levels and congestion,	
Deposit refund schemes and performance bonds	-	
Liability	-	
Tradable permits	Tradable efficiency permits for car manufacturers	
Voluntary agreements	-	
Information	-	

#### Soil degradation

Soil degradation		
Issue: reduce soil degradation from all causes but especially from water erosion		
Initiatives		
Fiscal incentives	Tax offsite damages	
	Subsidies to good practice	
Charges	-	
Deposit refund schemes and performance bonds	-	
Liability	-	
Tradable permits	-	
Voluntary agreements	Management agreements	
Information	Extension services	

#### **Structure for policy packages**

A policy package paper exists for each environmental problem. The structure of the policy packages is based on the following format:

- 1. *Key issues* associated with each environmental problem are described. This may include a comment about the expected benefits from environmental control as well as the most suited policies based on the five criteria;
- 2. *Available instruments*: based on the matrices that 'match' policies to environmental problems presented above and guided by the results of the 'policy assessment', the policy packages give more detail to the recommend policies. The section also provides *experience with policy instruments* in the EU15 (and elsewhere if relevant) and where possible an indication of the *effectiveness of the policy instruments* is given, i.e. a summary of what is known about the effectiveness of actual instruments, including simulations of hypothetical instruments and judgements.

#### Structure of policy assessments

This section assesses the suggested policies against the five criteria described above, i.e.

- i) causal criterion,
- ii) efficiency criterion,
- iii) administrative complexity,
- iv) equity criterion, and
- v) jurisdictional criterion.

## **3. MONETARY VALUATION TECHNIQUES**

#### Introduction

The economic approach to valuing environmental changes is based on people's preferences for changes in the state of their environment. Environmental resources typically provide goods and services for which there are either no apparent markets or very imperfect markets, but which nevertheless can be important influences on people's well-being. Examples include the quality of air, which affects people's health, crop yields, damage to buildings, and acidification of forests and fresh waters.

However, the lack of markets for these services means that unlike man-made products, they are not priced, therefore their monetary values to people cannot be readily observed. The underlying principle for economic valuation of environmental resources, just as for man-made products, is that people's *willingness to pay* (WTP) for an environmental benefit, or conversely, their *willingness to accept compensation* (WTA) for environmental degradation, is the appropriate basis for valuation.

If these quantities can be measured, then economic valuation allows environmental impacts to be compared on the same basis as financial costs and benefits of the different scenarios for environmental pollution control. This then permits an evaluation of the net social costs and benefits of each scenario for the different environmental issue.

The lack of markets and prices for many environmental goods and services means that the challenge for economists is twofold. The first task is to *identify* the ways in which an environmental change affects well-being; this is addressed in the next section, where the components of 'total economic value' of a resource are explained. The second task is to *estimate the value of these changes* through a variety of direct and indirect valuation techniques, exposition of which is given in the following sections.

#### **Total Economic Value**

The monetary measure of the change in society's well-being due to a change in environmental assets or quality is called the total economic value (TEV) of the change. To account for the fact that a given environmental resource provides a variety of services to society, TEV can be disaggregated to consider the effects of changes on all aspects of well-being influenced by the existence of the resource.

TEV can be divided into *use values* and *non-use values*, the latter also being called 'passive use values'. Use values include:

- direct use values, where individuals make actual use of a resource for either commercial purposes (e.g. harvesting timber from a forest) or recreation (e.g. swimming in a lake)
- indirect use values, where society benefits from ecosystem functions (for example, watershed protection or carbon sequestration by forests)
- option values, where individuals are willing to pay for the option of using a resource in the future (for example, future visits to a wilderness area)

Non-use values can take the form of:

- existence values, which reflect the fact that people value resources for 'moral' or 'altruistic' reasons, unrelated to current or future use
- bequest values, which measure people's willingness to pay to ensure their heirs will be able to use a resource in the future

Typically it is not possible to separate existence and bequest values.

To arrive at an estimate of the net change in societal well-being arising from an environmental change, we must consider each of these elements in turn. The total economic value (TEV) of a change is the sum of both use and non-use values:

TEV = use values + non-use values

= direct use + indirect use + option + existence + bequest values

Table 2.1 presents a taxonomy for environmental resource valuation, using the total economic value of a forest as an illustration.

Total Economic Value				
	Use Values		Non-use	Values
Direct Use	Indirect Use	Option Value	Bequest Value	Existence Value
Outputs directly consumable	Functional benefits	Future direct and indirect values	Use and non-use value of environmental legacy	value from knowledge of continued existence
<ul> <li>food</li> <li>biomass</li> <li>recreation</li> <li>health</li> </ul>	<ul> <li>flood control</li> <li>storm protection</li> <li>nutrient cycles</li> </ul>	<ul> <li>biodiversity</li> <li>conserved habitats</li> </ul>	<ul> <li>habitats</li> <li>prevention of irreversible change</li> </ul>	<ul> <li>habitats</li> <li>species</li> <li>genetic</li> <li>ecosystem</li> </ul>

 Table 2.1
 Economic taxonomy for environmental resource valuation

The first step in estimating any of these values is the definition and measurement of the environmental problem. This often includes an element of scientific uncertainty that can, at times, be quite significant. The accuracy of economic valuation is therefore dependent on accurate scientific identification and quantification of the environmental change in order to estimate people's preferences for or against it.

#### Valuation Techniques

The practical problem with economic valuation is one of deriving credible estimates of people's values in contexts where there are either no apparent markets, or very imperfect markets. In the case of marketed goods, price is the measure of willingness to pay and can be readily observed. However, in the case of non-marketed goods and services we need to elicit this value in different ways. There are two broad approaches to valuation, each comprising several different techniques, as illustrated in Figure 2.1.

- **Revealed preference techniques,** which infer preferences from actual, observed, market-based information. Preferences for environmental goods are revealed indirectly when individuals purchase marketed goods which are related to the environmental good in some way.
- Stated preference techniques, which attempt to elicit preferences directly by use of questionnaire, such as contingent valuation. All valuation of non-use values depends on these techniques.

We consider each of these approaches in turn, highlighting when each could be used, their advantages and drawbacks and their applicability to waste management problems.



Figure 2.1: A typology of monetary valuation methods

Total Economic Value

#### **Revealed Preference Techniques**

The essence of revealed preference techniques is that they infer environmental values from markets in which environmental factors have an influence. For example, there are markets for certain goods to which environmental commodities are related, as either substitutes or complements to the goods in question. In this way people's actions in actual markets reflect, to a certain extent, their preferences for environmental assets.

There are four main revealed preference techniques that are considered in the sections that follow.

- 1. Averting behaviour
- 2. Hedonic pricing (of property and labour)
- 3. Travel cost method
- 4. Random utility and discrete choice modelling

#### **Averting Behaviour**

The basis for the averting behaviour technique is the observation that marketed goods can act as substitutes for environmental goods in certain circumstances. When a decline in environmental quality occurs, expenditures can be made to mitigate the effects and protect the household from welfare reductions. For instance, expenditure on sound insulation can indicate households' valuation of noise reduction; expenditure on household water filters can be used to estimate economic values of clean water.

The method is applicable in situations where households spend money to offset environmental impacts. It requires data on the environmental change and its associated substitution effects. Fairly crude approximations can be found by simply looking directly at changes in expenditures on the substitute good resulting from some environmental change.

Advantages of these models are that they have relatively modest data requirements and can provide theoretically sound estimates based on actual expenditures. However, they can give incorrect estimates if other important aspects of individuals' behavioural responses are ignored. For example, individuals may engage in more than one form of averting behaviour in response to any one environmental change. Additionally, the averting behaviour may have other beneficial effects that are not considered explicitly, for example sound insulation may also reduce heat loss from a home. Furthermore, averting behaviour is often not a continuous decision but a discrete one: for example, a smoke alarm is either purchased or not. In this case the technique will tend to underestimate the value of the environmental good.

#### **Hedonic Pricing**

This technique depends on analysis of existing markets where environmental factors have an influence on price. The example most frequently used is that of the housing market, as the environmental attributes of a property will vary according to its location. For example, noise levels will be higher close to an airport and, other characteristics being equal, this can be expected to lower the price of a property in the area. Similarly, two identical properties which differ only in, say, the local air quality, will differ in value to the extent that people find one air quality preferable to the other. The difference can be viewed as the value attached to the difference in air quality as measured by willingness to pay (WTP).

The hedonic property price (HPP) method can be used even when properties differ in many factors other than environmental quality provided that data are detailed enough. With the use of appropriate statistical techniques, the hedonic approach attempts to (i) identify how much

of a price differential is due to a particular environmental difference between properties, and (ii) infer how much people are willing to pay for an improvement in environmental quality that they face and what the social value of the improvement is. The same technique has also been applied to labour in the valuation of work-related risk in hedonic wage (HW) studies. Identification of wage differentials due to differences in safety risks, for example, will give an indication of willingness to accept compensation (WTA) for incurring these risks, which can be used as a measure of the benefits of improving safety.

This technique can in theory be used to estimate factors such as the disamenity costs of location near to landfill sites, or air quality near to incinerators.

#### **Travel Cost Method**

Many natural resources are used extensively for the purpose of recreation. It is often difficult, however, to value these resources because no prices generally exist for them. The travel cost approach is based on the fact that, in many cases, a trip to a recreational site requires an individual to incur costs in terms of travel, entry fees, on-site expenditures and time. These costs of consuming the services of the environmental asset are used as a proxy for the value of the recreation site and changes in its quality.

Clearly, because travel cost models are concerned with active participation they measure only the use value associated with any recreation site. The method is now well -established as a technique for valuing the non-market benefits of outdoor recreation resources. It is useful because it is based on actual observed behaviour. However, the technical and data requirements are such that it is not readily applicable.

#### **Random Utility or Discrete Choice Models**

While the travel cost method is useful for measuring total demand or WTP for a recreational site, this technique is less useful for estimating the value of particular features or assets of the site which may be of interest. Random utility models have been developed for this purpose.

The emphasis of random utility or 'discrete choice' models is on explaining the choice between two or more goods with varying environmental attributes as a function of their characteristics. This can be useful where, for example, polluting activity causes damage to some features of a recreational site but leaves others relatively unharmed.

This can be illustrated using a simple example from a choice of transport mode. Supposing that, when undertaking a given journey, an individual faces the choice of travelling by taxi or by public transport. A taxi will take 20 minutes and cost  $\in$ 5, whereas public transport will take an hour but cost  $\in$  2. If the individual chooses to travel by taxi, it can be inferred that s/he judges the difference of 40 minutes in time to be worth at least the  $\in$  3 difference in fare. In other words, the value of the individual's time is at least  $\notin$ 4.50 per hour.

Another example is the choice between bottled water and tap water for drinking. The former is more expensive but associated with better quality. Therefore, the price difference between bottled and tap water is an indication of the value of risk in this context.

#### **Replacement Cost**

The replacement cost technique uses the cost of replacing or restoring a damaged asset to its original state as the measure of the benefit of restoration. The approach is widely used, largely because it is comparatively easy to apply.

This approach is valid where it is possible to argue that the remedial work must take place because of some other constraint such as an environmental standard. Replacement will only be efficient, however, if the environmental standard itself is economically efficient. Otherwise Information on replacement costs can be obtained from direct observation of actual expenditure on restoring damaged assets or from engineering estimates of restoration costs. The technique implies various assumptions, for instance, that complete replacement is, in fact, feasible. In general, however, the potential for confusion between costs and benefits means that the replacement cost technique needs to be applied with some care.

#### **Stated Preference Techniques**

Stated preference techniques enable economic values to be estimated for a wide range of commodities which are not traded in markets. In addition, these techniques are the only way to estimate non-use value of environmental resources. Here, we consider two approaches:

- 1. Contingent valuation
- 2. Conjoint analysis

#### **Contingent Valuation**

In contingent valuation (CV) studies, people are asked directly to state what they are willing to pay for a benefit or to avoid a cost, or, conversely, what they are willing to accept to forego a benefit or tolerate a cost. A contingent market defines the good itself, the institutional context in which it would be provided, and the way it would be financed. The situation the respondent is asked to value is hypothetical (hence, 'contingent') although respondents are assumed to behave as though they were in a real market. Structured questions and various forms of 'bidding game' can be devised to assess the maximum willingness to pay. Econometric techniques are then applied to the survey results to derive the average bid value, i.e. the average WTP.

There are three basic parts to most CV surveys. First, a hypothetical description of the terms under which the good or service is to be offered is presented to the respondent. Information is provided on the quality and reliability of provision, timing and logistics, and the method of payment. Second, the respondent is asked questions to determine how much s/he would value a good or service if confronted with the opportunity to obtain it under the specified terms and conditions. These questions take the form of asking how much an individual is willing to pay for some change in provision. Respondents are reminded of the need to make compensating adjustments in other types of expenditure to accommodate this additional financial transaction. Econometric models are then used to infer WTP for or WTA the change. Finally, questions about the socio-economic and demographic characteristics of the respondent are asked in order to relate the answers respondents give to the valuation question to other characteristics of the respondent, and to those of the policy-relevant population.

CV is likely to be most reliable for valuing environmental gains, particularly when familiar goods are considered, such as local recreational amenities. While the accuracy of results also depends on careful construction of the survey, a set of guidelines for applying CV to derive reliable estimates of non-use values is developed by the US National Oceanic and Atmospheric Administration (NOAA) panel (Arrow et al., 1993). This is now being extended to cover all CV studies.

#### **Conjoint Analysis**

Conjoint analysis (CA) is a broad term used to cover several different techniques, all of which are survey methods, but they involve asking individuals to rank alternatives rather than explicitly express a WTP or WTA. For contingent ranking, the inclusion of prices in some of the alternatives enables rankings to be converted to monetary values. Other aspects are similar to contingent valuation.

Again, the main application of relevance to the current study has been in the context of human health and landscape effects, as well as disamenity.

#### **Dose- and Exposure-Response Functions**

Dose-response functions (DRFs) measure the relationship between a unit concentration of a pollutant and its impact on the relevant receptor. Exposure-response functions (ERFs) are based on the same principle but measure the response with respect to the exposure. Exposure is a measure of the levels of a pollutant in the environment surrounding the receptor in question. For example, a person may be exposed to a certain concentration of an atmospheric pollutant, but the dose received will depend on the amount inhaled, which is higher during exercise and lower during rest. In general, effects will be more closely related to dose, but it is much easier to measure exposure. Hence it is important to recognise that any dose-response function is often represented by the approximation of an exposure-response function (ApSimon et. Al., 1997).

Dose-response techniques are used extensively where a physical relationship between some cause of damage, such as pollution, and an environmental impact or 'response' is known and can be measured. Once the relationship has been estimated, then WTP measures derived from either conventional market prices (which are adjusted if markets are not efficient) or revealed/inferred prices (where no markets exist) using one of the techniques described in the previous section. The physical damage is multiplied by this shadow price, or value per unit of physical damage, to give a 'monetary damage function'.

The approach is theoretically sound, and can be used wherever the physical and ecological relationships between a pollutant and its output or impact are known. The specification of the D/ERF is crucial to the accuracy of this technique, and is the main source of uncertainty. Difficulties and uncertainties may arise in: identifying the pollutant responsible for the damage and all possible variables affected; isolating the effects of different causes to determine the impact on a receptor, e.g. synergistic effects where several pollutants or sources exist; identification of damage threshold levels and the long term effects of low to medium levels of pollution. All these problems make it difficult to determine the appropriate empirical specification of the functional form. Additionally, there is the further complication that evidence of a physical response may not be economically relevant if individuals are not concerned about it and, therefore, do not attach a value to avoiding it. For these reasons, large quantities of data may be required and the approach may be costly to undertake.

If, however, the D/ERFs already exist and the impacts are marginal, the method can be very inexpensive and provide reasonable first approximations to the true economic value measures.

## 4. BENEFITS TRANSFER

The benefit assessment procedure conducted in this study makes extensive use of 'benefits transfer', i.e. taking existing monetary valuation (willingness to pay) studies and applying them outside of the site contexts where the study was originally carried out. There is in fact no alternative to this procedure if any use at all is to be made of benefit valuation techniques. The approach implicitly underlies the procedures used, for example, by ExternE, although, as it happens, their use of transferred functions may be more basic than ours in at least one respect. They choose specific functions and apply these across Europe. In our case we make some attempt to engage in 'meta studies' where that is possible. Technically, the alternative is to carry out willingness to pay studies across all EU15 countries for all environmental problems. Clearly, this is not possible. Nonetheless, we should be aware that the procedure involves risks and errors. This note serves to set out the nature and problems involved in benefits transfer. It should be noted that the literature analysing the validity of transfer techniques is very small.

#### (a) Transferring average WTP from a single study to another site which has no study

The basic idea is to 'borrow' an estimate of WTP in context i and apply it to context j, but making adjustments for the different features of the two contexts. For example, if incomes vary we might have

#### $WTP_i = WTP_i(Y_i/Y_i)^e$

where Y is income per capita, WTP is willingness to pay, and 'e' is the income elasticity of demand, i is usually called the *study* site and j the *policy* site.

A typical example of such an approach is given by Krupnick et al., (1996) who transfer US WTP for various health states to Eastern Europe using the ratio of wages in the two areas and an income elasticity of demand of 0.035. The significance of the procedure can be realised since the wage ratio raised to e=0.035 produces a WTP in Eastern Europe equal to only 8% of that in the USA.

A second, common adjustment is for population size and, less frequently, for the distribution of population characteristics, e.g. age.

Note that the transfer is 'assumed' to be correct: no separate validation is carried out. This is similar to most of the transfer of values used in the EU Priorities study.

#### (b) Testing the equality of means at two sites where studies exist

Where there are two sites both with actual WTP estimates we can obtain some idea of the validity of benefits transfers by comparing the two mean WTPs. We wish to know if they are statistically the same. If they are, then there is some reason to feel confident that the results from a given site can be transferred to another site, as in (a) above.

Where the underlying distribution of WTP is thought to be normal, parametric tests can be used (eg t-tests) to determine if the mean WTP results at the two (or more) sites are statistically the same. Where this restriction is thought to be unreasonable, then non-parametric tests are required. More sophisticated testing can be done, e.g. to find out if the two underlying WTP distribution (not just the means) are statistically the same.

and

#### (c) Transferring benefit functions

A more sophisticated approach is to transfer the *benefit function* from i and apply it to j. Thus if we know that WTPi = F(A, B, C, Y) where A,B,C are factors affecting WTP at site i, then we can estimate WTPj using the coefficients from this equation but using the values of A, B, C, Y at site j.

Alternatively, we can use *meta analysis* to take the results from a number of studies and analyse them in such a way that the variations in WTP found in those studies can be explained. This should enable better transfer of values since we can find out what WTP depends on. Whole functions are transferred rather than average values, but the functions do not come from the single site i, but from a collections of studies.

#### (d) Is transferring functions valid?

How do we know if transferring functions is a valid procedure? As with the procedure under (a), we have no direct test that the result is 'correct'. The literature has proceeded by taking estimated demand functions at site i and site j and then comparing them to see if, statistically, they are the same. This involves at least testing for the equivalence of the coefficients in the two functions, e.g.

$$WTP_i = x + a_1 A + b_1 B + c_1 C$$

$$WTP_j = x + a_2 A + b_2 B + c_2 C$$

so that we require  $a_1 = a_2$  etc, where equality here is statistical equality (Loomis, 1992).

Recent literature has suggested that even if it is valid to transfer *benefit functions*, based on statistical equality of coefficients, the resulting estimates of *benefits* may be in error. This is because benefits may not be a linear function of the coefficients. Downing and Ozuna (1996) take demand functions for 8 sites in Texas and conclude that around 50% of functions are transferable (have the same coefficients) but that only a small minority would yield reliable benefit estimates. This has led Bergland et al., (1995) to suggest that both valuation functions and benefits estimates must be transferable (see the 'protocol' below..)

Generally, the literature testifies to the unreliability of transferring benefit functions (Loomis, 1992; Downing and Ozuna, 1996; Bergland et al., 1995; Parsons and Kealy, 1994). Most studies seem to suggest that transferring functions is better than transferring average values, but that both are subject to significant margins of error (Kirchhoff et al., 1997).

#### (e) Validating benefits transfer

The test in (d) above involves taking actual demand functions and seeing whether they are statistically the same and will produce similar benefit estimates. Another test would be to take a WTP estimate from i and apply it to j using a simple procedure such as the one set out in (a) above. Then, a full WTP study would be carried out in j and the mean WTP result would be compared with the 'transferred' WTP.

Navrud (1997) has done this for minor impaired health states to see if WTP estimates from the USA can be transferred to Europe (in fact, to Norway). He concludes that the transferred estimates significantly overstate the 'actual' WTP as derived from a contingent valuation study in Norway.

Alberini et al., (1995) make this test of benefits transfer using two US contingent valuation studies of a 'restricted activity day' due to a head cold and transferring the results to Taiwan. In this case the transfer multiplier was (Yj/Yi) which implies e=1.

They then carried out a contingent valuation of the morbidity effect in Taiwan. The results were statistically the same, ie the simple benefits transfer approach accurately predicts the policy site study results.

#### (f) The Bergland-Magnussen-Navrud Protocol

Bergland et al., (1995) (BMN) recommend testing for benefits transfer in four stages:

- 1 test that mean WTP<sub>i</sub> = WTP<sub>j</sub>, using parametric and non-parametric tests depending on the assumed underlying distribution of WTP
- 2 estimate WTP'<sub>j</sub> where WTP'<sub>j</sub> uses estimated parameters from i and the actual values of explanatory variables at j. Test for the equivalence of WTP'<sub>j</sub> = WTP<sub>j</sub>, ie we require

WTP'<sub>i</sub> = 
$$f(b_i, X_j) = WTP_j$$

and correspondingly for WTP'i.

3 compare parameters b in each study, with the requirement that

$$b'_i = b_j$$
 and

 $b'_j = b_i$ 

where  $b'_{j}$  comes from estimating the function WTP'\_{j} = f(b\_{i}, X\_{j}) above, and correspondingly for  $b'_{i}$ .

4 test for the proposition that the two benefit functions come from one underlying function with parameters b such that

$$b = b_i = b_i$$
.

Criteria for successful Benefits Transfer

It appears generally agreed that successful benefits transfer requires:

- 1 adequate data for those studies included in the analysis
- 2 sound economic and statistical technique
- 3 studies with regressions of WTP on determining variables
- 4 similar populations in the compared sites
- 5 similarity of the environmental good to be valued
- 6 similar sites
- 7 similar distributions of property rights.

See, for example, Brouwer and Spaninks (1997).

#### Conclusions

The literature on benefits transfer is very small. The attractions of benefits transfer are very clear: without it, one has to resort to primary valuation studies. This is both expensive and time consuming. It would not matter for 'micro' problems where it is often possible to carry out such studies, but it is a problem for wide-ranging studies such as the European Environmental Priorities study where we require valuations across many Environmental problems, across the EU 15 countries and, where possible, across Accession countries.

At the moment, the literature reports mixed results with the balance of opinion expressing considerable caution about benefits transfer. It seems clear that the conditions required for 'good' transfer are not met in the kind of the analysis where single estimates are applied across man countries. The error is likely to be reduced substantially wherever meta analysis can be done and meta-functions can be applied. Even here, there are some doubts about the validity of transfer.

# **5. INCOME ELASTICITY OF DEMAND FOR THE ENVIRONMENT**

#### Benefits Transfer and the Income Elasticity of Demand

Finding 'unit values' for changes in water quality or water availability rests heavily on benefits transfer, i.e. the process of taking values from one context and applying them to another. Benefits transfer usually involves adjustments to the original estimates, adjustments that can be quite complex but which are often very simple. A formula that is quite widely used but which adjusts only for differences in income between the original site (i) and the site to which the estimate is to be transferred (j) is:

$$B_i = B_{i} \cdot (Y_i / Y_i)^e$$
 ...[1]

where 'B' is benefit (measured by willingness to pay), 'Y' is income, and 'e' is the 'income elasticity of environmental value'.

Here we focus on how to find values of 'e'.

The Analytics

There are two parameters that are relevant in estimating the income elasticity of demand for the environment.

The first is the conventional measure of income elasticity of demand

 $\eta = \Delta X.Y/\Delta Y.X$  ...[2]

where,  $\Delta$  is change in, X is the quantity of the environmental good in question and Y is income.

Value of η	Share of expenditure on good X as Y rises	Name of good	
η <0	Falls	inferior	
$0 < \eta > 1$	Falls	normal, necessity	
$\eta = 1$	Constant	normal	
$\eta > 1$	Rises	normal, luxury	

Traditionally, goods have been classified in terms of the value of  $\eta$ :

Casual commentators have often argued that the environment is a *luxury good*, i.e. that it is something that societies worry about only when incomes rise.

The second indicator is the *income elasticity of environmental valuation*, or willingness to pay (WTP):

$$e = \Delta WTP.Y/\Delta Y.WTP$$
 ...[3]

The relationship between [2] and [3] is easily derived. From [3] we have

$$Y/\Delta Y = e.X/\Delta X$$

and substituting in [3] gives:

#### $e = \eta \Delta WTP X / WTP \Delta X$ [4]

Note that e < 1 is quite compatible with  $\eta > 1$ , so that a good that is a 'luxury good' can have an income elasticity of WTP < 1 (Flores and Carson, 1997).

Which is the relevant concept ? Since the focus of most environmental policy is on public goods that have some quantity constraint, it turns out that it is the second concept - the income elasticity of environmental values - that is more relevant.

#### Empirical evidence (a) environmental goods

In an early survey, Pearce (1980) assembled what evidence there was on income elasticities of WTP for the environment. He found (a) that hedonic property price models could not be used to *infer* income elasticities because the models themselves tended to constrain the values to be unity anyway, and (b) what evidence there was suggested that income elasticities were *less than unity*. Kristrom and Riera (1996) review more recent evidence and reach a similar conclusion. Analysing six European contingent valuation studies, Kristrom and Riera find that the share of expenditure on environment falls as income rises, i.e. e < 1. This result is supported with evidence from Australian and US CVM studies, with other work from Africa and the USA, and with analysis of corporate donations to environmental causes in the USA.

#### Empirical evidence (b) risks

Values of e < 1 have also been obtained from studies of valuations of statistical life. These produce a range of values of 0.3 to 1.1 with the majority of estimates being at the lower end of the range, ie well below unity:

Blomquist, 1979	e = 0.3
Jones-Lee et al., 1985	e = 0.3
Persson and Cedervall, 1991	e = 0.3
Jones-Lee et al., 1993	e = 0.3
Miller and Guria, 1991	e = 0.3 to 0.6
Persson et al., 1995	e = 0.46
Viscusi and Evans, 1990	e = 1.0 (non fatal injuries)
Kidholm, 1994	e = 1.1
(taken from a survey by NERA, 1997).	

While the number of studies remains limited, it is difficult to avoid the conclusion that the environment is *not* a luxury good on the relevant definition of income elasticity of WTP.
# 6. VALUING STATISTICAL LIVES

# Introduction: the Importance of Risk Valuation in Environmental Cost-Benefit Studies

Environmental cost-benefit studies include as benefits any reductions in the risks of premature mortality and morbidity. In turn, changes in the risks of health 'end points' are given economic valuations based on the willingness to pay (WTP) of those at risk to reduce the risks. Valuations may vary with the level of risk and certainly vary with the health state that is avoided, e.g. people are more averse to cancer risks than risks of accident. One feature of these cost benefit studies is that health benefits tend to dominate overall benefit estimates. Accordingly, if the basis on which the health benefits are estimated is incorrect, then the overall cost-benefit result is very likely to be incorrect. It matters a great deal, therefore, if the underlying epidemiology is correct and if the economic valuation applied to the health effects is correct.

This paper provides an overview of the issues as they relate to premature mortality only. It is designed as a background paper on the debate about the appropriate way to treat life risks in the context of environmental change. It does not seek to produce any new results, being designed mainly for reference and as a guide to the issues.

Table 5.1 shows the role that health benefit valuation has played in some recent European costbenefit studies. It can be seen that the overall benefits figures are dominated by health impacts. Other studies report cost-benefit results for policies aimed directly at health effects. Here the issue is whether benefits exceed costs, an issue that is also very much affected by the approach taken to health impacts.

Study	Title and subject area	Health benefits as % of total benefits
Holland and Krewitt, 1996	Benefits of an Acidification Strategy for the European Union: reductions of SO <sub>x</sub> , NO <sub>x</sub> , NH <sub>3</sub> in the European Union	86-94%. Total benefits cover health, crops and materials.
AEA Technology, 1998a	<i>Cost Benefit Analysis of Proposals</i> <i>Under the UNECE Multi-Effect</i> <i>Protocol</i> : reductions of SO <sub>x</sub> , NO <sub>x</sub> , NH <sub>3</sub> , VOCs	80-93%. Total benefits cover health, crops, buildings, forests, ecosystems, visibility
IVM, NILU and IIASA, 1997	Economic Evaluation of Air Quality for Sulphur Dioxide, Nitrogen Dioxide, Fine and Suspended Particulate Matter and Lead: reductions of these pollutants	32-98%. Total benefits include health and materials damage
AEA Technology, 1998b	Economic Evaluation of the Control of Acidification and Ground Level Ozone: reductions of $NO_x$ and VOCs. $SO_2$ and $NH_4$ held constant.	52-85% depending on inclusion or not of chronic health benefits. Total benefits include health, crops, materials and visibility

 Table 5.
 Health benefits as a percentage of overall benefits in recent cost-benefit studies

# **Economic Valuation and Resource Allocation**

Economic valuation is intricately inseparable from the issue of how to allocate scarce resources. Risk reduction is not a costless activity and hence any resources used up in the reduction of one set of risks could have been used to reduce another set of risks. Taking a wider view, resources allocated to risk reduction might equally be allocated to some entirely different purpose: education, restoring national heritage, improving landscapes, and so on. Valuation attempts to provide the answer to the problem of choosing between alternative uses of resources. If risk reduction has a high value relative to other uses of resources, then risk reduction should have priority.

There have been philosophical objections to the use of economic valuation. On what might be called the 'rights approach' individuals have rights to human health and a clean environment, and such rights would have similar status to rights against discrimination (Bullard, 1994). One possible implication of the rights approach is that all environmental risks should be reduced to zero, since any positive level of risk infringes individuals' rights. Alternatively, if rights conflict and are not absolute, then some trade-off between rights has to occur. A variation of the rights-based approach extends rights to non-human species, i.e. it confers 'value' on living things, and sometimes non-living things independently of human values. This is sometimes articulated in terms of 'intrinsic' rights of species to exist.

The rights-based approach contrasts with the view based on 'trade offs' between cost and risk reduction. Risk reduction is pursued up to some point where the costs of such action are thought to be 'too high'. There are divergent views as to how this trade-off is to be made. In particular, there are those who favour a balancing of economically valued costs and benefits, and there are those who favour leaving the trade-off to the political system. This categorisation is not meant to be all -encompassing. More detail of the considerable variation of views within these categories can be found in Turner (1993).

One of the problems with the debate about these alternative views is that much of the discussion takes place quite independently of the real world context of environmental change. If resources were infinite there would be no problem of trade-off, and hence no problem of determining priorities. Everything deemed to be 'good' or 'right' could be done. But the real world is not like this and it is necessary to choose. The fundamental feature of choice making is cost, which is another way of saying that resources are finite. Adopting a rights-based approach implies that the choices surrendered by pursuing risk reduction as a matter of right (i.e. the cost) are of a lower 'moral order' than risks to human health or risks to other species. The problem then is that risk reduction has to be pursued regardless of the forgone values sacrificed. Moreover, all risk reduction has to be pursued: it cannot be correct to reduce some risks but not others unless the rights are attenuated in some way. Risk reduction may therefore conflict with other rights, e.g. rights to a decent livelihood, rights to education, and, especially, rights to freedom of choice.

Much of the motive for the rights-based approach arises from an understandable sense of frustration with the fact that trade-off approaches do involve 'acceptance' of some positive levels of risk. But it also has its foundations in a lack of appreciation of what 'cost' actually means, a perception fostered by the view that cost is 'just money', as if money is unrepresentative of human wellbeing.

Developments in risk analysis sharply underscore the unavoidability of trade-offs and the unreality of the rights-based approach. Risk-risk analysis and health-health analysis draw attention to the fact that the costs of risk reduction policies are met from reductions in household incomes - see Keeney (1990, 1994, 1997), Graham et al., (1992), Lutter and Morrall (1994), Portney and Stavins (1994), Viscusi (1994) and Viscusi and Zeckhauser (1994). It is known that households with low incomes tend to have higher exposure generally to life and health risks, so that reductions in expenditure increase risk exposure. For example, Keeney (1997) estimates that in the USA there is one fatality for each \$5-11 million of public expenditure on risk reduction.

Finally, rights-based approaches tend to be discussed as if whatever is deemed to be 'right' by one or more persons constrains others who may not share the moral view. Put another way, what is right has an absolutist flavour. If there were no dispute about the moral standpoint, then, clearly, there would be a moral consensus. But in so far as hypothetical market studies have shown the existence of 'lexical' preferences (implying no trade off) - and this is disputed - they

have not characterised the whole sample. It is unclear therefore what role a minority believing in 'rights' should play in determining the outcome of a policy or project choice.

Overall, then, rights-based approaches fail because of their neglect of the most basic of all economic principles - opportunity cost, and because they have little to say about consensus.

#### The Trade -Off View: Economics

The economic approach to the trade-off issue operates through the aggregation of human preferences. The set of persons affected by a decision defines the set of people whose preferences count, where 'affected by' means that their wellbeing is, in one way or the other, partly dependent on the environment in question. This preference-base is inherently 'democratic' - it requires that policies be responsive to preferences however they are formed. Preferences are revealed in the market -place through demand behaviour -i.e. as 'willingness to pay' (WTP). Indeed, the demand curve in textbook economics is a (marginal) willingness to pay curve. If WTP is rejected as a criterion for allocating resources to risk reduction, then some explanation has to be provided as to why environmental goods and services are different to other goods and services which are allocated on a WTP basis.

But risk reduction often has no market, i.e. the issue giving rise to risk is not bought and sold on the open market. Clean air would be an example. Thus the economic approach requires that preferences for risk reduction be inferred from human behaviour in other contexts.

The theory of economic valuation has developed substantially in the last two decades. This section reviews, briefly, those techniques that relate to human health risks only. Other techniques are relevant to the valuation of other environmental changes. For a detailed review see Freeman (1993).

For a change in risk that threatens life and health generally, we can say that the relevant valuation is the value that the individual at risk attached to their own health and life chances, plus what others would be willing to pay to avoid the risk to that individual, plus any costs that society at large bears and which would not otherwise occur if the individual did not suffer the effects of the risk in question. These components of this value of risk (VOR) are:

(a)	VOR <sub>i,i</sub>	where i,i refers to the individual i's valuation of risk to themselves, i.e. 'own risk'. The way in which these individual VORs are aggregated is dealt with shortly. Essentially, we will require the summation of such own valuations for all individuals at risk to give $\Sigma_i VOR_{i,i}$ , more commonly known as the 'value of a statistical life' - see below.
(b)	VOR <sub>i,j</sub>	where the i,j notation now refers to j's valuation of risks to i. Again, this will need to be summed for all j, i.e. for all people expressing some concern about risks to i, to give $\Sigma_j$ VOR <sub>i,j</sub> .
(c)	COI <sub>i</sub>	where COI refers to the 'cost of illness' suffered by i but which costs are borne by the rest of society. An example would be hospital costs. COI could be regarded as part of VORi,j

The extent to which these three components of the value of life risks can in fact be *aggregated* is discussed later.

#### Valuing Statistical Lives

One form of health risk is the risk of premature mortality arising from some risk context, say increased air pollution. What value should be attached to such risks of mortality? The sum of individuals' own valuations of risks to their own lives is known as the value of a statistical life, VOSL. The shorthand often used for the VOSL is 'value of life', which is unfortunate. Since the idea of 'valuing life' appears odd to some and morally offensive to others, it is important to understand what a value of a statistical life (VOSL) actually is.

The way a VOSL is obtained is by aggregating up from a value (willingness to pay, WTP) of risk reduction. Imagine the probability of dying next year is 0.004 for each person and suppose we have 1000 persons in the population. Assume there is some risk reduction policy that reduces the risk to 0.003, a change of 0.001. Each person is asked to express their WTP for this change in risk and suppose the answer is £1000. The risk reduction policy is a public good: it affects everyone equally. Thus 1000 people say they are each willing to pay £1000 for the policy, i.e. their aggregate willingness to pay is £1 million. The change in risk will result in one statistical person being saved each year (1000 x 0.001). Thus the value of a statistical life is £1 million in this example. It is important to understand that no one is being asked their WTP to avoid themselves dying at a specified time: they are being asked to express a WTP for a change in risk. As Freeman (1993) notes:

"...the economic question being dealt with here is not about how much an individual would be willing to pay to avoid his or her certain death or how much compensation that individual would require to accept that death. In this respect, the term "value of life" is an unfortunate phrase that does not reflect the true nature of the question at hand. Most people would be willing to pay their total wealth to avoid certain death; and there is probably no finite sum of money that could compensate an individual for the sure loss of life. Rather, the economic question is about how much the individual would be willing to pay to achieve a small reduction in the probability of death during a given period or how much compensation that individual would require to accept a small increase in that probability." (p320).

It is worth emphasising Freeman's point: the VOSL is *not* what someone is willing to pay to avoid losing their life, a confusion that is pervasive in the popular literature commenting on valuations of life risks. It is the valuation of small changes in risk. VOSL is essentially a convenient rule for aggregation.

Individuals' WTP to reduce risks can be expected to vary across different individuals. The two main reasons for this will be that:

(a) people have differing attitudes to risk: some may even be 'risk lovers', i.e. positively enjoying risky contexts. Most people are risk avoiders, i.e. they will tend to reveal a positive willingness to pay for risk reduction. But there is no particular reason why their valuations of risk should be the same;

(b) incomes vary and hence willingness to pay is likely to vary in such a way that those with higher incomes have higher WTPs. This is not a necessary result since attitudes to risk may vary in such a way as to offset an income effect. Nonetheless, it raises an important equity issue about fairness between people, an issue that is not in fact confined to risk valuations but to the use of WTP measures in general.

A VOSL can also be measured by a 'willingness to accept' compensation for increased risk. It is well known that many people do make this trade-off between risk and money, for example by accepting premia on wages to tolerate risk. It is tempting to think that the WTA approach will produce very much higher values for a VOSL than the WTP approach, simply because WTA is not constrained by income. WTP and WTA can, indeed, be different and WTA for environmental losses may exceed WTP for environmental gains by factors of 2-5 (Gregory, 1986). Various explanations exist for this disparity, including the fact that individuals may feel they are losing an 'entitlement' if the issue is one of loss of an entitlement (WTA) rather than an increment to an existing entitlement (WTP). Another explanation, which is wholly consistent with economic theory, suggests that WTA > WTP arises mainly in contexts where there is no ready substitute for the environmental good in question (Hanemann, 1991). These issues are discussed further later on.

#### **Techniques for Estimating VOSL**

A number of techniques have been developed to estimate VOSLs. The main ones are rooted in the general economic theory of valuation, i.e. they have a theoretical basis on the measurement of human wellbeing based on individuals' preferences. One widely used technique, however, has only a tenuous link to the theory.

#### Valuing Mortality Risks: Wage Risk Models

The wage risk, or 'hedonic wage' model estimates a willingness to accept measure of risk. Essentially, it looks at wages in risky occupations and seeks to determine the factors that determine wages. One of these factors is hypothesised to be the risk level. Other things being equal, workers will prefer jobs with less risk to jobs with high risk. This will result in a relative shortage of workers for risky jobs and hence wages in those jobs should be higher. This 'wage premium' then becomes a measure of risk valuation. It can be estimated by multiple regression techniques in which the wage is the dependent variable and the various factors influencing the wage are the independent variables. An example might be:

where Educ is education, Exp is years of experience, Union is an indicator of the degree of unionisation of the labour force, Risk is the objective (or perceived) probability of fatal injury and Occ is some indicator of the desirability of the occupation. The coefficient linking Wage and Risk is then the WTA measure of risk.

One obvious problem with such approaches is that workers have to know about the differences in risks and, if they do, whether those perceptions coincide with 'objective' measures of risk such as the probability of a fatality in that industry. If there is no perception of risk, but risk exists, then the 'hedonic wage' (i.e. the wage premium) may be zero, seriously understating risk values. If there is a perception of risk but it is exaggerated compared to objective risk, then risk may be overvalued. Other problems include the potential for workers in risky jobs to be 'self-selecting', i.e. those tolerant of risk may be attracted into the industry in question. Lack of labour mobility will also mean that some workers will remain in jobs without full compensation for the risks involved.

What limited evidence there is suggests that workers actually overstate the risk of their jobs. But as Freeman (1993) points out, what matters for the hedonic wage model is the perception of differences in risks between jobs, not the absolute level of risk in a given job.

Most hedonic wage studies have been carried out in the USA and suggest that VOSLs range from \$2 million (1994 prices) to \$3.5 million.

Individuals spend money in trying to reduce risks, so called 'averting behaviour'. Under certain circumstances these expenditures approximate the economist's concept of WTP to reduce risk. The kinds of averting expenditures in question might be on smoke alarms; safety harnesses, tamper-proof drug storage containers, and so on. These kinds of expenditures can be regarded as part of what is called a 'health production function' in which the state of good health is 'produced' by various factors, including expenditures on averting ill-health. Note that some apparent averting expenditures are not valid measures of risk reduction. Thus, it is quite widely assumed that life insurance expenditures are measures of WTP to avoid risk. But insurance expenditures do not have the effect of reducing risks. Indeed, they may actually increase risks by encouraging less careful behaviour - the issue of 'moral hazard'. As Freeman (1993) notes, life insurance essentially values the earning capacity of the insured individual to the dependants who are the ones who will gain from any insurance policy. This is not at all the same thing as the individual's willingness to pay to reduce risks to his or her own life, which is what is required.

The health production function can be written:

H = f(Poll, Med, Avert, Other)

where Poll is the level of pollution, Med is the level of medical treatment, and Avert is the level of averting activity. 'Other' refers to all the other factors affecting health status: age, income, smoking behaviour, and so on. Reducing pollution will reduce the time spent being unwell, say from 4 days to 3 days. If by spending £X through avertive behaviour the same reduction in ill-health can be achieved, then £X should be the value of the reduced pollution to the individual. More formally,

WTP (Pollution Reduction) = (Reduced Time in Ill Health) x (Extra Cost of 'Producing' Health by Mitigating Activity).

In this way, expenditures on risk reduction can be interpreted as WTP for risk reduction. In practice, finding examples of avertive expenditures that are 'purely' health producing has proved difficult. Studies include seat belt use and smoke detectors and suggest VOSLs of about \$0.7 million and \$2.2 million.

#### Valuing Mortality Risk: Contingent Valuation

The contingent valuation method (CVM) requires that individuals express their preferences in response to a questionnaire. It is therefore very much akin to market research in which the researcher seeks to find out how a respondent would behave, in terms of WTP, for a modified or new good. Questionnaires take two forms: (a) open-ended or continuous approaches simply ask what someone is WTP (or WTA), and no prompting of likely values is permitted; and (b) discrete or dichotomous choice in which the value is posed and the respondent is then asked whether he or she is willing to pay that sum, yes or no. Yes/no questions that use the cost of providing some project or benefit as the sum to which the yes/no answer is sought are also known as 'referendum' approaches. There is now a general preference for the dichotomous choice format. The kinds of biases that may occur in CVM include:

(a) Starting point bias in the dichotomous choice format, i.e. respondents tend to produce WTP answers that tend towards the first 'price' put forward by the questioner. Such a bias is easily tested by seeing if the difference between the average stated WTP is statistically different to the starting point sum;

- (b) Strategic bias whereby the respondent understates the true value of their preference (they 'free ride') in the expectation that others will state more and thus secure the good in question for everyone. This phenomenon was long thought to be inherent with 'public goods', such as clean air, since if the clean air is provided for any one individual it is provided for everyone (clean air is said to be 'jointly consumed'). Of course, the misstatement of preferences may be biased the other way: someone may be so keen to see the good provided that they overstate their preferences, fearing that others will free ride. Tests for strategic bias suggest that, contrary to expectation, it may not be of major significance. Such tests may involve stratifying the sample deliberately to give some people strong incentives to free ride and others less of an incentive, and then seeing if their average WTPs diverge. Others involve indicating that unless more than a certain percentage of respondents vote for the good, it will not be supplied.
- (c) Hypothetical bias: the respondent may produce answers that are purely hypothetical, i.e. if the good or policy in question is actually provided, their WTP will be less than stated in response to the questionnaire. Careful design of questionnaires can reduce the hypothetical bias problem to very low levels in WTP questionnaires. It may be more of a problem with WTA questions. Measuring the bias usually involves comparing stated preferences with actual preferences when real sums of money are involved in the CVM. Since respondents are less familiar with compensation contexts (the relevant context for WTA), their stated WTA is likely to exceed their actual WTA.
- (d) Part-whole or 'mental account' bias: here the problem is that, while the questionnaire may focus on a specific environmental benefit, the respondent may act as if he or she is valuing environmental improvement in general. Tests for this kind of bias involve varying the quantity of the good in question, e.g. a 1% reduction in risk, a 10% reduction in risk and so on, and seeing whether WTP varies significantly as the benefit is increased. A number of studies suggest that WTP is the same and that what individuals are 'purchasing' is not the benefit in question but the 'warm glow of giving' or 'moral satisfaction'. One response has been to ensure that questions are not asked until the respondent has been reminded that he or she has a specific budget to be allocated. The empirical evidence of part-whole bias remains mixed.
- (e) Information bias: the quality of information supplied to the respondent may affect the stated WTPs. Usually, the more information is supplied about the risks in question the higher the WTP. Why this is regarded by some critics of CVM as a flaw in the method is unclear. Information should influence WTP in exactly the same way as information influences WTP in the everyday market place for ordinary goods. While there is an interesting issue of how much information should be provided, what matters most is that the same level of information be provided to all respondents.

There are other problems with the CVM but modern CVM design is capable of minimising the extent of error in stated responses. Good surveys of CVM are to be found in Pearce et al., (1994) and Bateman and Turner (1993).

# Valuing Mortality Risk: The Human Capital Approach

Before the formalisation of hedonic wage, CVM and avertive behaviour approaches the most commonly used technique to value risk to human life was the human capital approach. The idea is simple: an individual is 'worth' to society what he or she would have produced in the remainder of their lifetime, gross of taxes since the interest is in society's valuation of the individual. An argument did exist as to whether the earnings that are relevant should be net or gross of the individual's own consumption. If the individual's consumption is excluded then the value concept is simply that of how the rest of society values the individual, and that is inconsistent with the WTP approach. If the individual's consumption is included, then at least

some gesture is made towards included a value from the individual's own standpoint. But there is in fact nothing to suggest that an individual's WTP need be equal to the remaining lifetime income of that individual. The link to the WTP approach is clearly tenuous at least. The WTP approach is based on how individuals value risks to their own lives, whereas the human capital approach makes no obvious reference to that concept. Indeed, the human capital approach says little about individuals' attitudes to risk. Its one virtue is that it is thought to be very easy to calculate (Rowlatt et al., 1998).

Even if the human capital approach is accepted as a rule of thumb, however, there are problems in its estimation. First, if the individual at risk has retired from work, the human capital valuation would suggest that their VOSL is zero or even negative (if consumption is deducted). This perhaps under lines the failure of the concept in terms of its theoretical underpinnings. If human capital was a WTP concept, those out of work should have positive WTPs. Second, someone may be of median age but still not be producing marketed 'output'. A house person, for example, produces non-market output and this would have to be valued. Third, future earnings cannot simply be added up year by year to get a total since the individual will discount the future. Hence a discount rate is needed.

While it would be better if the human capital approach was avoided altogether, it is still widely used, no doubt because of the relative ease of computation. If used, the computation in question would be:

'VOSL' =  $\sum_{i=1, T-t} (p_{t+i}, Y_{t+i})/(1+r)^{i}$ 

where  $\Sigma_{i=1,T-t}$  denotes the sum over time from time t, the current age of the individual at risk, T is the age at which the individual ceases to work,  $p_{t+i}$  is the probability of the individual surviving from age t to age t+i, Y is income, and r is the discount rate.

The human capital approach does not produce a value of statistical life in the sense of  $\Sigma_i$  VORi,i above. But can it be used to estimate the value that others put on the life at risk? For close relatives, friend etc. the answer must be 'no': such people do not value risks in terms of the income forgone by the individual at risk. But what of society generally? There is a sense in which society loses the output of the individual less the consumption of that individual. But had the individual survived, the rest of society might have to produce transfer payments in the form of welfare payments, and illness costs arising from the ill-health that the individual would have suffered had they survived. Arguably, then, what the rest of society loses is i's income minus i's consumption minus i's claims on the rest of society over the expected lifetime of i without the risky event. This is perhaps the most that can be said for the human capital approach.

# Willingness to Pay versus Willingness to Accept?

The valuation of statistical lives rests on the WTP or WTA principle. It is easy to see that WTP is constrained by income and/or wealth. WTA appears not to be so constrained since it is an amount in compensation for accepting a risk. Both the contingent valuation and wage-risk models have the capability to elicit WTA estimates and practice has found that WTA figures are not infinite, i.e. people do not expect extremely large payments in compensation for losses of environmental quality. But WTA does tend to exceed WTP, as noted earlier, and these differences cannot generally be explained by issues of questionnaire design. There are genuine differences between WTP and WTA. This raises the issue of which measure is correct?

The answer depends on the context, and especially on property rights, although other factors help to explain the size of the discrepancy between WTP and WTA. The matrix below explains the property rights issue. Generally, WTP is the right concept when the individual whose valuation is being sought does not have a right to the improvement being valued. WTA is the correct concept

when the individual does have a right to the status quo and is being asked to forgo a benefit or accept a loss.

Valuation of a GAIN	Valuation of a LOSS	
WILLINGNESS TO PAY =>	WILLINGNESS TO PAY TO AVOID THE LOSS =>	
Property rights DO NOT rest with individual	Property rights DO NOT rest with the individual	
Valuation of a GAIN	Valuation of a LOSS	
WILLINGNESS TO ACCEPT COMPENSATION TO FORGO THE GAIN =>	WILLINGNESS TO ACCEPT COMPENSATION =>	
Property rights DO rest with the individual	Property rights DO rest with the individual	

The relevance of property rights arises in two contexts. The first is where there are clearly defined legal rights, and the second relates to the individual's perception of rights. Thus, we might expect individuals to value a unit loss much more highly than a unit gain if he or she believes they have some right to the existing amount of environmental quality or asset. There is indeed evidence that individuals have 'loss aversion', i.e. they regard the status quo as some kind of reference point from which gains and losses are evaluated. This view is stressed by advocates of 'prospect theory' - see Kahneman and Tversky (1979).

A second factor explaining the wide divergence sometimes found between WTP and WTA is the degree of substitutability of the thing being valued with other goods. Suppose, for example, that what is being valued is a unique environmental or material asset - the Grand Canyon or the Taj Mahal, say. Then, as there are no ready substitutes one might expect WTA to be very much higher than WTP. And this turns out to be the case: the fewer the substitutes the larger the discrepancy between WTA and WTP, as theory would predict (Hanemann, 1991). Hanemann's explanation is not comprehensive because the same WTA/WTP discrepancy exists for commonplace goods, in which case the insights from prospect theory appear to be relevant.

The relevance to life risks is of course significant if individuals feel that risks to life or health constitute an invasion of their 'rights' not to have to tolerate those risks. There is some evidence to suggest that those rights will be especially pronounced when the risks are not voluntary, in contrast with, say, occupational risks. If so, we might expect wage risk models, which are WTA estimates for voluntary risk, to reveal risk valuations that are above, but not substantially above, WTP valuations. CVM models, on the other hand, might reveal substantial WTA/WTP discrepancies if the risk in question is involuntary. The problem with the evidence is that most of the VOSL studies are either wage-risk studies or CVM studies of transport risk. Transport risks may or may not be seen as voluntary compared to, say, radiation risks from a nuclear power plant, although most risk studies appear to treat transport-related risks as involuntary. Early analysis of the voluntary/involuntary risk valuation issue was fairly inconclusive. Starr (1972) attempted a comparison of risk levels and the associated benefits and concluded that involuntary risk might be valued by a factor of ten more than voluntary risk. Reworked by Otway and Cohen (1975), Starr's ratio appears far too high and a factor of two appears more appropriate. But further analysis by Fischoff et al., (1979) reinstates the large tenfold differential between involuntary and voluntary risk values. Substantial question marks hang over these studies however not because of the risk data but because of the use of measures of benefit based on actual expenditures on the activity in question or the contribution the activity makes to an individual's income. There is some affinity here with the required benefit measure - WTP - but it is far from precise.

Overall, then, the conceptual contexts in which WTP and WTA should be used are fairly clear and relate to the presumption about property rights. In practice, determining the assignment of property rights is far less straightforward.

#### The Evidence on VOSL

Several reviews exist of VOSLs.

Pearce, Bann and Georgiou (1992) review the various estimates of VOSL and find the mean estimates across studies shown in Table 5.2 (updated to 1997 values). The estimates also show the ratio of WTA to WTP because of the presumption that WTA studies tend to find higher values than WTP studies. The wage risk studies are fairly consistent between the UK and USA with a suggestion that a higher (average) value exists for WTA in the USA than in the UK. On the other hand, WTP studies appear to produce higher values in the UK. These data also suggest that the WTA > WTP inequality holds for the USA but not for the UK, but there is no ready explanation for these disparate results. Note, however, that the estimates shown are unweighted averages, i.e. it assumed that all the studies reviewed are equally valid.

Table 5.2Values of Statistical Life

UK£m (1997)	USA	UK	
WTA (wage risk)	2.9-4.6	2.4-2.9	
WTP (CVM, CRM)	1.2-2.2	3.3-5.3	
WTP (market)	0.9-1.0	0.5-2.8	
WTA/WTP (WR/CVM)	1.3-3.8	0.4-0.9	
WTA/WTP (WR/mkt)	2.9-5.1	0.9-5.8	

Source: Pearce et al. (1992) updated to 1997 prices

Other reviews for the USA suggest ranges of recommended VOSLs. Fisher *et al.*, (1989) recommend a range of \$2-10 million; Cropper and Freeman (1991) recommend \$2-6 million; Viscusi (1992) recommends \$3-7 million and Miller (1989) recommends \$1-4 million. An extensive review by Industrial Economics Incorporated (1993) fits a lognormal distribution to available estimates considered to be 'reliable' (26 in all) and takes the geometric mean (i.e. the mode) to obtain \$4 million in 1993 values, or \$4.5 million in 1997 values. Overall, then, VOSL estimates of around US\$ 1.6-4.8 million would appear to be 'safe'.

Use of VOSL estimates of the kind noted in Table 5.2 has come under criticism for several reasons:

- (a) There is unease about the fact that health benefits based on VOSL are so dominant in cost-benefit studies;
- (b) the VOSL estimates come largely from accident contexts where the mean age of the person killed is very much lower than in pollution contexts. There is therefore a feeling that older people, perhaps with an already impaired health state will not have the same valuation of risk as someone who is very much younger;

and

(c) it is, as noted above, very easy to confuse what a VOSL is actually measuring. Wrongly translated as a 'value of life,' the concept is easy prey for critics who do not invest in attempts to understand the analytical foundations of VOSL. Since this confusion is widespread, analysts often prefer not to use the VOSL concept at all.

Of these reasons, only the second has any intellectual basis, although the first does reflect a 'statistical sensitivity' issue in the sense that, if the VOSL estimates are wrong, then entire decisions may be changed.

For these good and bad reasons, then, there have been attempts to estimate not the value of the risk of fatality but the value of the life period gained by reducing the risk. This has come to be known as the 'value of a life year' or VOLY.

#### Values of Life Years (VOLYs)

The underlying rationale for valuing 'life years' is that many contexts in which health risks occur relate to pollution. Clearly, pollution is more likely to affect people who are most vulnerable. In a poor country this may be the very young and the very old. In a rich country, where infant mortality risks are very low, it is more likely to affect the elderly and especially those who are already at risk from their prevailing health state. Suppose, for argument's sake, that, statistically, the reduced life expectancy of someone exposed to air pollution is six months. Then, the argument goes, what matters is the value the individual places on those six months of extended life. If the period is a few weeks or even days, then the relevant value is that 'life period' rather than the actual risk. This contrasts with the VOSL where a person, however old they are, is faced with a risk and they express their WTP to reduce that risk. In principle, the two values - VOSL and VOLY - should bear some relationship since the person at risk must have some idea of remaining life expectancy. Indeed, it would be accounting for the remaining life period available to them.

One obvious way of approaching the problem is to see if WTP to reduce risks is functionally related to age, an issue we return to below. The surprising thing about the VOSL literature is that very little of it controls for age, so that only a few studies exist to offer a guide on how risk valuations vary with age.

Alternative approaches attempt to estimate the VOLY and, so far, two procedures have been used. The first simply takes estimates of the VOSL and converts them to values of life years; i.e. no additional information is sought. The second attempts to construct VOLYs from first principles by engaging in valuation studies that directly attempt to elicit the WTP for extended periods of life.

# VOLYs derived from VOSLs

One approach to estimating the VOLY is to regard it as the annuity which when discounted over the remaining life span of the individual at risk would equal the estimate of VOSL. Thus, if the VOSL of, say, £1.5 million relates to traffic accidents where the mean age of those involved in fatal accidents is such that the average remaining life expectancy would have been 40 years, then

where A = A(n,r) = 
$$\sum_{t=1}^{n} 1/(1+r)^{t}$$
 or A =  $[1-(1+r)^{-n}]/r$ .

and n is years of expected life remaining and r is the utility discount rate<sup>6</sup>. Examples are shown below for n = 40 years.

<sup>&</sup>lt;sup>6</sup> The utility discount rate is the rate at which future wellbeing is discounted, not the rate at which income or consumption is discounted. The UK Treasury (1997) adopts a rate of pure utility discounting of 1.5% but little evidence exists to support this rate. Pearce and Ulph (1995) suggest a rate of 0.3%.

VOSL	VOLY at r=0.3%	VOLY at r=1%	VOLY at r=1.5%
£m 1997 prices	£	£	£
	A = 37.6	A = 32.8	A = 29.9
1.0	26,595	30,460	33,445
1.5	39,894	45,690	50,167
2.0	53,190	60,920	66,890
3.0	79,787	91,138	100,000

r is utility discount rate.

These VOLY numbers can then be used to produce a revised VOSL allowing for age. At age 60, for example, suppose life expectancy is 75 years. The VOSL(60) is then given by

revised VOSL(60) = 
$$VOLY * \sum_{t=1}^{T-a} 1/(1+r)^{t}$$

where T-a = 15 is remaining life expectancy. In the case indicated, this would be, at 1% discount rate and a 'standard' VOSL of £1 million:

*revised* VOSL(60) = 
$$(30,460).(13.87) = \pounds 422,480.$$

The result is that the age-related VOSL declines with age and this appears to accord with the intuition of some commentators (see the discussion below). The generalised formula for age related VOSL is:

revised VOSL(a) = [VOSL(n)/A] \* 
$$\sum_{t=1}^{T-a} 1/(1+r)^{t}$$
 or  
revised VOSL(a) = VOSL(n) \* A(T-a,r)/A(T,r)

where a is the age of the individual or group at risk, T is life expectancy for that group, VOSL(a) is the age-adjusted VOSL and VOSL(n) is the 'normal' VOSL.

One advantage claimed for this approach to valuation is that it can be combined with other information on the health state of the individual at risk. This might be done via 'QALYs' -quality of life year ratings. QALYs involve weighting life expectation by quality factors that reflect individuals' own perceptions of the quality of life associated with that life expectancy. Extending a life by one year but with an associated level of pain and suffering thought to be unbearable would attract a low QALY indicator. A VOLY multiplied by this QALY would give a revised quality-adjusted VOLY (Davies and Teasdale, 1994).

While the VOLY approach may appear sound it suffers from a number of deficiencies.

First, it offers no evidence that VOSL declines with age in the manner shown. If this were to be the case, we would expect to find evidence that the WTP to reduce risks varies inversely with age. As Rowlatt et al., (1998) note, there *is* some evidence for a declining WTP as people become older, but that evidence is not at all consistent with the age profile of VOSL as dictated by the VOLY approach. Ignoring any influence from health states, the VOLY approach implies a monotonically declining VOSL with age, whereas the WTP for risk literature tends to produce inverted 'U' shapes. In essence, the age-related VOSLs derived on this approach are arbitrary: they are imposed from outside rather than being derived from any individual-based risk assessment. Maddison (1998) suggests that there are sound reasons for supposing that VOSL is

proportional to the number of discounted life years remaining to an individual and that it is inversely proportional to the survival probability in the current time period. In other words, Maddison suggests that there are rationales for a declining VOSL with age, but that this will be attenuated in old people by the reduced survival probability. For the UK, he suggests that the VOSL for a 74-year-old with six months life expectancy would be 17% of the healthy 36-year-old.

Second, while the evidence on age and WTP for risk reduction is not compelling, what there is suggests a decline in WTP. Jones-Lee (1989, 1993) reports WTP for accident reductions in the UK and these are shown in Figure 6.1. For illustration, they are compared there to the implied VOSLs that would come from using the VOLY approach. Notice that the VOLY-based VOSLs do not exhibit the 'inverted U' shape found in the Jones-Lee studies and they seriously understate later age VOSLs when compared to the standard VOSL approach. Also, the VOSL ratios using the VOLY approach are invariant with the value of VOSL(n), but will change with the discount rate assumed. However, Figure 6.1 shows that the VOLY-based VOSL is largely unaffected by the choice of 1% or 2% utility discount rates. Supporting evidence for modest declines in WTP with age can be found in Maier et al., (1989), Miller and Guria (1991), Kidholm (1995), Persson et al., (1995) and Desaigues and Rabl (1995). Rowlatt et al., (1998) cite a Swedish paper - Persson and Cedervall (1991) - which found rising values of WTP with age, a result that Rowlatt et al., put down to problems in eliciting answers to questions about small risk changes, but which could be consistent with theory (see below). Johannesson and Johansson (1996) also find modestly increasing WTP with age.



Figure 6.1 Value of Statistical Life (VOSL) as function of age according to Rawlett et al, 1998, Jones-Lee et al., 1989 and 1993, and for the VOLY-based VOSL approach for utility discount rates of 1% and 2% respectively.

Third, it was noted that the VOLY-based VOSL could be combined with QALY information. Again, it appears that the VOLY approach imposes an apparently 'logical form' on the valuations by assuming those already ill will value remaining life periods less. But to quote a recent study for the US Environment Protection Agency: "...it is possible that the reduced life expectancy and reduced enjoyment of life associated with many chronic illnesses may result in lower WTP to reduce risks of death. On the other hand, facing serious illness and reduced life expectancy may result in higher value [being] placed on protecting the remaining time." (Chestnut and Patterson, 1994).

Overall, Maddison's approach holds out some promise for finding age-related VOSLs via indirect routes. These should then be tested against VOSLs derived from direct approaches in which age is specifically accounted for.

#### VOLYs derived from WTP experiments

An alternative procedure based on the VOLY concept is to see the WTP to extend a lifetime conditional on having reached a certain age. Johannesson and Johansson (1996) report a contingent valuation study in Sweden where adults are asked their WTP for a new medical programme or technology that would extend expected lifetimes conditional on having reached the age of 75. Respondents are told that on reaching 75 they can expected to live for another 10 years. They are then asked their WTP to increase lifetimes by 11 years beyond 75, i.e. the 'value' of one extra year. The results suggest average WTP across the age groups of slightly less than 10,000 SEK using standard estimation procedures and 4,000 SEK using a more conservative approach. In dollar terms this is \$600-1500<sup>7</sup>. Recall that this is for one year of expected life increase. WTP actually *increases* with age, although not dramatically - on the standard basis, 8000 SEK for the 18-34 age group, 10,000 for the 35-51 age group and 11700 for the 51-69 age group. Using the formula:

VOSL(a) = 
$$VOLY * \sum_{t=1}^{T-a} 1/(1+r)^{t}$$

Johannesson and Johansson suggest these values are consistent with 'normal' VOSLs of \$30,000 to \$110,000, substantially less than the VOSLs derived previously. Since *T-a* is obviously less the older the age group, then the relevant VOSLs will decline with age. They also derive discount rates of 0.3% to 3.4% and these are invariant with age. Finally, they argue that these lower valuations are consistent with findings in Sweden and the USA on social attitudes to allocating resources to life saving. Thus, Cropper et al., (1994) found that survey respondents strongly favoured life saving programmes which save the lives of young people rather than old people. Earlier work by Johannesson and Johansson (1995a, 1995b) found that Swedish attitudes were similar, and that expectations about the future quality of life at old age play a significant role (regardless of what the actual quality of life is). The implications of the low WTP values for health care are hinted at in Johannesson and Johansson (1996): they observe that the VOSL values are 'negligible' compared to the costs of health treatment for the aged.

The Johannesson and Johannsson study is the only one available at present which attempts to value of life year directly. Is the WTP approach used consistent with the VOSL approach ? It is arguable that the 'goods' being valued are quite different: VOSL studies value risk and the VOSL is simply an aggregation of those individual valuations of risk. The WTP for a life year is not explicitly a value of risk, but a value of extending a life year once the respondent is assumed to reach a particular age. The Johannesson and Johansson paper could be argued to be more relevant for pollution control policy if the benefits of that policy are thought to accrue mainly to the elderly.

#### VOSL and Age Again

7

As opposed to accidents, environmental risks are especially likely to affect the health of those already predisposed to illness, e.g. the elderly. Hence, it is important to know if the VOSL is likely to vary with age. At one extreme we could legitimately argue that we have no reason to

The range if reported as \$400-\$1500 in the original article but this looks like a misprint.

suppose this WTP will vary negatively with age. Indeed, older people may be all the more risk averse simply because the value of time itself is likely to increase the less there is of it remaining to the person at risk. Plausible reasons to suppose that WTP will fall as age increases have been advanced in the theoretical literature (e.g. Freeman, 1993, Chapter 10; Cropper and Simon, 1994). Freeman (1993) reviews life cycle models and shows that, in general, one might expect WTP to decline with age. This is because lifetime utility is dependent on lifetime consumption in such models and older people simply have fewer consumption years left. However, there are several reasons why such life cycle WTP models understate 'true' WTP:

- (a) they tend to omit others' valuations of the life at risk (e.g. relatives, friends) (see below);
- (b) life cycle models assume that expected lifetime utility depends on expected lifetime consumption only, whereas individuals surely value survival as well. Note that this value of survival need not vary inversely with age at all, and could actually increase;
- (c) there is evidence to suggest that WTP for 'contemporaneous risk' is less than WTP for 'latent risk', i.e. WTP for avoiding accidents is less than WTP for avoiding risks of cancer (Jones-Lee *et al.*, (1985)). Yet the empirical VOSL literature is almost entirely based on accident risks. For pollution issues, then, transferring VOSLs from accident risk contexts to pollution contexts is likely to understate the 'true' degree of risk aversion.

#### Others' Valuation of Risks to an Individual

The second component of the basic valuation equation was the value placed on risks to i by others who are close to i, relatives and friends. The literature that seeks to estimate such valuations is very much smaller, but suggestive of some results. Viscusi et al., (1988) surveyed consumers to elicit risk valuations for injury risks from the use of insecticides in the USA. Consumers were asked their WTP to reduce risks from 15/10,000 to 10/10,000 for two pairs of risk: inhalation and skin poisoning and inhalation and child poisoning. The WTP figures of \$1.04 and \$1.84 respectively, therefore implies values of risk of \$2080 and \$3680 (1.04/0.0005 and 1.84/0.0005). Individuals were then asked their WTP for an advertising campaign to reduce risks by the same amount generally, i.e. to other people. The results implied valuations of the first risk pair of \$10,000 for North Carolina State - where the survey was conducted - and \$3,070 for risks outside the state. For the second risk pair, the values were \$18,100 and \$4,260. The state/non-state comparisons suggest that valuations decline as the individuals at risk become more 'anonymous' to the valuer, as one might expect.

An early study by Needleman (1976) sought the valuation of close relatives for reductions in risks. The study looked at kidney donors. Donors tended at that time to be close relatives to secure greater chances of acceptance of the transplanted organ. The kidney donor suffered a slight increase in risk while the recipient had dramatically improved chances of survival. By looking at data on actual kidney donations and at refusal rates - i.e. situations in which the relatives refused to make the donation - Needleman estimated a 'coefficient of concern'. An average coefficient of 0.46 implies that close relatives' valuations may be 46% of the value of risk of the individual at risk, i.e. one might write  $VOR_{i,j} = 0.46VOR_{i,i}$ , where j is now close relatives. Recall that VOR<sub>i,i</sub> is summed across all individuals at risk and expressing a positive WTP to obtain a VOSL. It follows that VOR<sub>ij</sub> should be summed across all close relatives of those at risk. The effect could be substantial. For example, if each individual at risk has four close relatives, the effect would be to multiply VOSL by  $4 \ge 0.46 = 1.64$  to obtain the summed valuations of close relatives. Schwab Christe and Soguel (1995) conduct a contingent valuation analysis of willingness to pay to avoid the consequences of a road accident. WTP was estimated in two contexts: where the respondent was the hypothetical victim and where the respondent is a relative of the hypothetical victim. In each case, the pain and suffering of others is relevant. In the former case, willingness to pay (VOR<sub>ii</sub> in our notation) may already account for the pain and suffering of relatives and others, i.e. WTP is influenced by the concern the victim has for the

effects of an accident to him/herself on others. In the second case, where the victim is a relative, WTP (VOR<sub>i,j</sub> in our notation) may reflect both the relative's own bereavement and also some judgement of the pain and suffering of the victim. Schwab Christe and Soguel try to distinguish these effects. The results are:

- (a) VOR<sub>i,i</sub> for a death is 1.7 million Swiss francs, or around 1.2 million US\$;
- (b) VOR<sub>i,i</sub> for an accident involving severe and permanent disability is slightly higher than VOR<sub>i,i</sub> for death at some 1.75 m Swiss francs;
- (c)  $VOR_{i,j}$  for relatives (j) is *higher* than VORi,i at around 2 million Swiss francs, and higher still for permanent and severe disablement. In general VORi,j would appear to be equal to 1.25  $VOR_{i,i}$ , about three times the effect found by Needleman's study.

Cropper and Sussman (1988) suggest that US citizens have a willingness to pay for children's statistical lives equal to 70-110% of their own values (VOR<sub>i,i</sub>). This is consistent with a New Zealand study by Miller and Guriua (1992) with a VOR<sub>i,j</sub> of 119% for family members. Blomquist *et al.*, (1996) estimate a VOR<sub>i,i</sub> of \$2 million and a VOR<sub>i,j</sub> for children by parents of \$3-5 million, i.e. 1.5-2.5 times the VOR<sub>i,i</sub>. Blomquist *et al.*, (1996) also review other studies of VOR<sub>i,j</sub>, finding a fairly consistent range of values between 23 and 50% of VOR<sub>i,i</sub> when the person at risk is not a family member.

The studies suggest that  $VOR_{i,j}$  may be of the order of 100% for own family members and perhaps 20% for non-family members. The implications of adding 20% premia for *each person* affected by the ith life at risk are fairly significant. Not only would a typical valuation of, say, \$2 million be quadrupled because of close family valuations, but a further \$0.4 million (20% of  $VOR_{i,i}$ ) might need to be added for each person thought to exhibit a degree of concern for the individual at risk. VOSLs, then, could be seriously understated by focusing on  $VOR_{i,i}$  alone.

However, the issue of aggregating life risks across individuals is complex. For a discussion see Johansson (1995). Jones-Lee (1992) cautions against assuming that  $VOR_{i,i}$  and  $VOR_{i,j}$  can be added but suggests a social value of a statistical life of 1.1 to 1.4 times the  $VOR_{i,i}$ . This is based on analysis of altruistic motives. For *pure altruism* - in which the person exhibiting the concern respects the preferences of the person at risk - the correct VOSL is the 'own' valuation. The original proof is given in Bergstrom (1982). Jones-Lee (1991) examines the case of *pure paternalism* - where j exhibits a concern for i's risks but does so on the basis of overriding i's preferences - and concludes that the same result holds, i.e.  $VOSL_{i,i}$  is the correct valuation. Where there is a focus by j on i's 'safety', i.e. risk reduction, and the utility function for j takes the form:

$$U_j = U(x_j, s_j; s_i)$$

where x is the private good and s is safety, then it is legitimate to add a 'premium' to the own VOSL. Thus, for any premium to be justified, j's preferences have to be paternalistic and relate only to i's safety, not to i's consumption of the private good.

# Valuing Future Lives

Given that 'sustainable development' is a widely embraced goal of economic and environmental policy, and given that 'sustainability' raises the importance of impacts on future generations, one issue of some importance in risk valuation is that of how to value 'future lives'. Essentially, should a life at risk in, say, 50 years time be valued in the same way as a life today ? This is an 'intergenerational equity' issue. Jones-Lee and Loomes (1993) have shown that, on balance, future lives should be valued at the current VOSL and should *not* be discounted. Or, put another way, the effective discount rate applied to future lives should be zero provided the valuations

$$VOSL_{t=50} = VOSL_0$$
,

the 'equal values no discounting' rule

or 
$$VOSL_{t=50} = VOSL_0.e^{50g}.e^{-50r}$$

the 'discounted future values' rule, where g = expected rate of income growth and r = the discount rate. So long as r=g the two rules are the same. The rules become more complex once we allow for the degree of aversion to inequality that might be displayed by the current generation; once a distinction is made between the discount factor for future risks and the discount factor for future income; and once survival probabilities vary between generations. In general:

- (a) the greater the degree of aversion to inequality, the closer one gets to the equal values and no discounting case;
- (b) the greater the survival probabilities of future generations relative to current generations the more justified is discounting future risk reduction benefits; and
- (c) only if future wellbeing (as opposed to income) is discounted, can discount rates greater than zero be justified in the context where the current VOSL is used to value future risks.

More generally, either future risks are valued at future WTP levels and then discounted in the same way as income, or future risks are valued at current VOSLs and no discounting is allowed, provided there is impartiality between current and future generations.

#### Valuing Statistical Lives When Incomes are Unequal

WTP and, less obviously, WTA estimates of VOSL are constrained by income. WTP and WTA estimates are also averages, i.e. there is a frequency distribution from which the mean is taken, so that some people have much higher valuations of risk than the mean and some have much lower valuations than the mean. One of the reasons for these different valuations will be income differences within the nation. This procedure for deriving a VOSL has given rise to extensive misunderstanding. Imagine two countries, one rich and one poor, such that the rich country imposes a risk on the poor country through pollution. Global warming, which results from the emission of greenhouse gases, is often regarded as an example of such 'imposed' pollution costs. (Although the rich world (the OECD countries) actually emits just under 40% of total greenhouse gases, with 60% coming from the developing world, oil rich nations, and the ex-Soviet Union (World Resources Institute, 1994)). Estimates of VOSLs determined by WTP estimates in the rich and poor countries will produce higher values for the rich country than the poor one, WTP being (partly) determined by income levels. Suppose the rich country's pollution gives rise to an estimated 100 premature mortalities in the poor country. Assume the rich country faces the choice of spending resources on international pollution control to the benefit of the poor countries, or spending the same level of resources on a domestic issue which also saves 100 lives, i.e. the marginal cost of saving lives is the same in the two countries. A cost-benefit test will result in the resources being spent domestically because the 100 'domestic' lives will be 'worth' more than the 100 overseas lives due to the higher risk valuations. Yet if 'all lives are equal' in some sense, such an outcome seems very unfair, especially if the rich country can be said to impose the pollution on the poor country. Some have argued that, if a VOSL is to be used

at all, it should be the same VOSL for everyone and that VOSL should be the higher of the two figures, i.e. the VOSL for the rich country.

Is the benefit-cost test then invalid in some moral sense? There are several issues to be distinguished.

First, the VOSL within a country is an average, as noted above. In principle, then, the same procedure should be used where VOSLs differ across countries. The resulting VOSL will be an average of the two VOSLs, but it will not be the highest VOSL that is used. If the highest figure was chosen, then, logically, it must also be chosen within a country, i.e. the average should not be used. Such an outcome is not logically tenable since the individual with the highest aversion to risk would then determine everyone's valuations. The idea of averaging valuations to reflect concern about the inequality of WTP is a long standing one in cost-benefit analysis. Pearce (1986, original edition 1971) discusses a rule in which WTP is weighted by a ratio of average income to actual income, i.e. an adjusted WTP for any country i becomes: WTPi\* = WTPi. $\Box$ /Yi

and WTPj is

WTPj\* = WTPj. $\Box$ /Yj

where  $\Box$  is the average of Yi and Yj. The ratio of the two WTPs is then

 $WTPi*/WTPj* = \frac{WTPi.Yj}{WTPj.Yi}$ 

This procedure will produce the same 'common VOSL value' if the value of risk as a proportion of income is the same in both i and j. Only if WTPj as a proportion of Yj is higher than WTPi/Yi will the resulting VOSL be higher in j than in i, and if the proportion is higher in i than in j, then the weighted VOSL will be higher in i than in j.

Second, regardless of the equity weighting procedure discussed above, the cost benefit test need not produce the unfair outcome discussed in the example above. This is because the example assumes a common marginal cost of reducing risks in both countries. In practice, risk reduction is likely to be less costly in the poorer country than in the rich country. Even with 'unequal lives' then, nothing follows about the outcome of a benefit-cost test.

#### Conclusions

Few topics have proved so controversial as the 'value of statistical life'. In large part the controversy derives from unfortunate terminology, since what appears to be at stake is the 'value of life' itself. This confusion has not been helped by even the most distinguished commentators and analysts using this phrase. But what is being estimated is the value of risk reduction. VOSLs are, essentially, convenient ways of aggregating these estimates.

In a finite world there really should be no dispute that resources have to be allocated rationally across different life risks. The real focus of the debate should be on the size of the VOSL. As we saw, this is the subject of a debate, which centres on two approaches to valuing risks. The first asks for the WTP to avoid risks, and the second asks for the WTP to extend an expected lifetime by some finite period, say one year. The literature on 'value of life years' turns out to be a hybrid of these approaches, deriving VOLYs from a given VOSL. As discussed, there appears to be limited theoretical justification for this hybrid approach. It is also not consistent with what we know about VOSLs as they vary with age. Nonetheless, what we know about the age-WTP relationship is not much. In turn, the literature that attempts directly to estimate VOLY is minute.

Such as it is, it suggests VOSLs are very much less than those derived from standardised VOR calculations.

Other issues concern the role that others' valuation of risks should play and the role that discounting might play in valuing future risks. In general it would appear that there is a case for adding a modest premium to own VOSLs for others' paternalistic concerns, and there is no strong case for discounting future risks.

# References

Aasnes. J., Bye. T., Mysen. H. T. 1996. Welfare effects of emission taxes in Norway, Energy Economics, 18, 335-346.

Action Asthma, 1990. The Occurrence and Cost of Asthma, Action Asthma, Cambridge Medical Publications, Worthing.

Acutt. M.Z., Dodgson. J.S. 1997. Controlling the Environmental Impacts of Transport: Matching Instruments to Objectives, Transportation Research D, 2(1), 17-33.

Adams. R., Crocker. T. 1991. Materials damages, in J Braden and C Kolstad (eds), *Measuring the Demand for Environmental Quality*, North Holland, Amsterdam.

Adger. N., Brown. K. 1994. Land Use and the Causes of Global Warming, Wiley, Chichester. AEA Technology, 1998a. Cost-Benefit Analysis of Proposals Under the UNECE Multi-Pollutant Multi-Effect Protocol, Report to UK Department of Environment, Transport and Regions, London and to UNECE Task Force on Economic Aspects of Abatement Strategies, Geneva.

AEA Technology, 1998b. *Economic Evaluation of the Control of Acidification and Ground Level Ozone*, (provisional), Report to DGXI of the European Commission, Brussels.

AEA Technology. 1998. *Economic Evaluation of Air Quality Targets for CO and Benzene*, for DGXI European Commission, Brussels.

AEA Technology. 1999. Economic evaluation of Proposals Under the UNECE Multi-effects and Multipollutant Protocol, report to European Commission, DGX1, (restricted).

Aimola. A. 1998. Individual WTPs for reductions in cancer death risks, in R Bishop and D Romano (eds), *Environmental Resource Valuation: Applications of the Contingent Valuation Method in Italy*, Kluwer, Dordrecht, 196-212.

Alberini, A., Cropper, M., Fu, T-T., Krupnick, A., Liu, J-T., Shaw, D., and Harrington, W. (1995). Valuing health effects of air pollution in developing countries: the case of Taiwan, *Discussion Paper 95-01: Resources for the Future*, Washington DC.

Alfsen. K., Brendemoan. A., Glomsrod. S. 1992. *Benefits of Climate Policies: Some Tentative Calculations*, Discussion Paper 69, Central Bureau of Statistics, Oslo.

Alfsen. K., Birkelund. H., Aaserud. M. 1995. Impacts of an EC Carbon/Energy and Deregulating Thermal Power Supply on CO2, SO2 and NOx Emissions, *Environment and Resource Economics*, **5**, 165-189.

Alfsen, K., Rosendahl, K. 1996. *Economic Damage of Air Pollution*, Statistics Norway, Oslo, *mimeo*. APHEA . 1996. Short Term Effects of Air Pollution on Health: a European Approach using Epidemiological Time Series Data, Journal of Epidemiology and Community Health, 50 Supplement 1, whole supplement.

Alfsen. K., Holtsmark. B., Torvanger. A. 1998. Trading greenhouse gas permits: some possible consequences of the Kyoto Protocol, Centre for International Climate and Environmental Research (CICERO), Working Paper 1998:2 (in Norwegian).

Applied Research Consultants, (ARC). 1997. Global Benefits and Costs of the Montreal Protocol on Substances that Deplete the Ozone Layer, reviewed by Environment Canada.

Apsimon. H., Cowell. R. 1995. *Estimating the Reduced Damage to Buildings in Europe due to Implementation of the Second Sulphur Protocol*, Energy Policy, July 1996.

Apsimon. H., Pearce. D., Ozdemiroglu. E. (eds). 1997. *Acid Rain in Europe: Counting the Cost*, Earthscan Publications, London.

Arden-Clarke. C., Evans. R. 1993. Soil erosion and conservation, in D Pimentel (ed), *World Soil Erosion and Conservation*, Cambridge University Press, Cambridge.

Ascari. S., Bernasconi. M. 1997. *The Economics of Risk and Uncertainty and the Valuation of Severe Accidents*, ExternE working document.

Audit Commission. 1997. '*Local Authority Performance Indicators*', Audit Commission, London. B.Ostro, 1995a. Addressing Uncertainties in the Quantitative Estimation of Air Pollution Health Effects, Paper presented to EC/IEA/OECD *Workshop on the External Costs of Energy*, Brussels, January 1995.

Bahn. O., Fragniere. E., Kypreos. S. 1998. Swiss Energy Taxation Options to Curb CO<sub>2</sub> Emissions, European Environment, 8, 94-101.

Bailey. A., Rapsomanikis. G. 1999. Producer response to the taxation of pesticide inputs in UK arable agriculture: a profit function approach, Agricultural Economics and Business Management, Wye College, *mimeo*.

Barker. T. 1993. Secondary Benefits of Greenhouse Gas Abatement - The Effects of a UK Carbon/Energy Tax on Air Pollution, Department of Applied Economics, Cambridge University, *Mimeo*.

Barker. T., Baylis. S., Bryden. C. 1994. Achieving the Rio Target: CO<sub>2</sub> Abatement Through Fiscal Policy in the UK, Energy-Environment-Economy Modelling, Discussion Paper No. 9, Dept of Applied Economics, Cambridge, 1994.

Baron. R. 1999. Carbon and energy taxes in OECD countries, in J Hacker and A Pelchen (eds), Goals *and Economic Instruments for the Achievement of Global Warming Mitigation in Europe*, Kluwer, Dordrecht, 207-22.

Bate. R. N., Dubourg. W. R. 1997. A Netback Analysis of Irrigation Water Demand in East Anglia, Journal of Environmental Management, Volume 49, 311-322.

Bateman, I. and Turner, R.K. 1993. The Contingent Valuation Method, in R.K.Turner (ed), *Sustainable Environmental Economics and Management*, Belhaven Press, London.

Bateman. I., Willis. K., Garrod. G. 1994. Consistency between Contingent Valuation Estimates: a Comparison of Two Studies of UK National Parks, *Regional Studies*, 28.5, 457-474.

Bateman. I., Brainard. J., Lovett. A. 1995. Modelling Woodland Recreation Demand using Geographical Information Systems: a Benefit Transfer Study, CSERGE working Paper GEC 95/06.

Bateman. I., Langford. I., Graham. A. 1995. A Survey of Non-Users Willingness to Pay to prevent Saline Flooding in the Norfolk Broads, CSERGE working Paper GEC 95/11.

Bateman. I., Diamand. E., Langford. I.H., Jones. A. 1996. Household Willingness to Pay and Farmers' Willingness to Accept Compensation for Establishing a Recreational Woodland, *Journal of Environmental Planning and management*, 39 (1), 21-43.

Beaumont. P. 1995. *Pesticide Reduction: Economic Instruments*, Worldwide Fund for Nature, Gland, Switzerland.

Bergland, O., Magnussen, K., and Navrud, S. (1995). Benefits transfer: testing for accuracy and reliability, Discussion Paper 03/1995, Department of Economics, Agricultural University of Norway, *mimeo*.

Bergstrom, T. 1982. When is a Man's Life Worth More Than his Human Capital ?, in M.Jones-Lee (ed), *The Value of Life and Safety*, North Holland, Amsterdam, 3-26.

Bertrand. N. 1997. *Meta-Analysis of Studies into Willingness to Pay to Reduce Traffic Noise*, CSERGE, University College London, London.

Bishop, R. 1978. Endangered Species and Uncertainty: the Economics of a Safe Minimum Standard, *American Journal of Agricultural Economics*, 60, 10-18.

Blomquist, G., Miller, T. and Levy, D. 1996. Values of Risk Reduction Implied by Motorist use of Protection Equipment, *Journal of Transport Economics and Policy*, January, 55-66.

Bolt. C. 1993. *The Cost of Quality: Establishing Customers' Willingness to Pay in a Regulated Monopoly*, OFWAT, Birmingham.

Boman. M. 1995. Estimating Costs and Genetic Benefits of Various Sizes of Predator Populations: The Case of Bear, Wolf, Wolverine and Lynx in Sweden, *Journal of Environmental Management*, 43, 349-357.

Bonnieux. F., Rainelli. P. 1994. *Learning from the Amoco Cadiz oil spill: damage valuation and court's ruling*, Fondazione Eni Enrico Mattei, Nota di Lavoro 18.94, Milan.

Bos. E., Vu. M., Massiah. E., Bulatao. R. 1994. *World Population Projections 1994-1995 Edition*, Johns Hopkins University Press, Baltimore.

Bowers, J. 1993. A Conspectus on Valuing the Environment, *Journal of Environmental Planning and Management*, Vol.36, No.1, 91-100.

Brack. D. 1996. International Trade and the Montreal Protocol, Earthscan, London.

Brisson. A., Pearce. D.W. 1995. *Benefits Transfer for Disamenity from Waste Disposal*, Centre for Social and Economic Research on the Global Environment, CSERGE, University College London, Working Paper 95-06.

Brouwer. R., Baidu-Forson. J., Sivakmar. M.V.K. 1995. Case studies of eco-regional approaches at the farm/enterprise level: response. Kluwer Academic. 401-08.

Brouwer, R., and Spaninks, F. 1997. Benefits transfer and the economic valuation of environmental damage in the European Union with special reference to health, CSERGE, University of East Anglia and University College London, *mimeo*.

Brouwer. R., Spaninks. F. 1997. The Validity of Transferring Environmental Benefits: Further Empirical Testing, CSERGE working Paper GEC 97/07.

Brown, M., Cox .R., Bull, K., Dyke, H., Sanders, G., Fowler, D., Smith, R., Ashmore, M. 1998. Brunekreef, B. 1995. *Quantifying the Health Effects for the Netherlands of Exposure to PM*<sub>10</sub>, National Institute of Public Health and Environmental Protection, Bilthoven, Report 623710002 (In Dutch).

Bruvoll. A. 1998. Taxing virgin materials: an approach to waste problems, *Resources, Conservation and Recycling*, 22, 15-29.

Bruvoll. A., Bye. T. 1998. Methane emissions and permit prices for greenhouse gases, *Economic Survey*, 4/98 Central Bureau of Statistics, Oslo.

Bullard, R.D. 1994. Unequal Environmental Protection: Incorporating Environmental Justice in Decision Making, in A.Finkel and D Golding, *Worst Things First: the Debate over Risk Based National Environmental Priorities*, Resources for the Future Inc, Washington DC., 237-266.

Bullock. C., Kay. J. 1997. Preservation and Change in the Upland Landscape: the Public Benefits of Grazing Management, *Journal of Environmental Planning and Management*, 40 (3), 315-334.

Burtraw. D., Toman. M. 1997. *The Benefits of Reduced Air Pollution in the US From Greenhouse Gas Mitigation Policies*, Discussion Paper 98-01, Resources for the Future, Washington DC.

Calthrop. E., Pearce. D. 1997. Methodologies for calculating the damage from air pollution to buildings and materials: An overview, in V.Kucera, D.Pearce and Y-W Brodin (eds) *Economic evaluation of air pollution damage to materials*, Swedish environmental protection agency 148-161.

Cambridge Econometrics. 1996. Competitiveness and the Carbon Tax, Report to UK Department of the Environment, London.

Cansier. D., Krumm. R. 1997. *Air pollution Taxation: An Empirical Survey*, Economics Dept, University of T∫bingen.

Capros. P. 1999. Evaluation of Macroeconomic Implications of Environmental Scenarios, Report to RIVM, 'Economic Priorities for Environmental Plan', EC DGX1.

Carson. R., Mitchell. R., Hanemann. W. M., Kopp. R., Presser. S., Ruud. P. 1994. Contingent valuation and lost passive use: damages from the Exxon Valdez, *Resources for the Future Discussion Paper 94-18*, Resources for the Future, Washington DC.

Carson, R., Mitchell, R., Hanemann, M., Kopp, R., Presser, S. and Ruud, P. 1995. *Contingent Valuation and Lost Passive Use Damages from the Exxon Valdez*, Discussion Paper 95-02, Department of Economics, University of California, San Diego, *mimeo*.

Carver, P.H. and Boland, J.J.1980. *Short and long-run effects of price on municipal water use*, Water Resources Research, Volume 16, No 4, August.

CBD (Convention on Biological Diversity). 1997. Recommendations for a Core Set of Indicators of Biological Diversity, UNEP/CBD/SBSTTA/3/9.

CEC. 1976. Council Directive of 8 December 1975 concerning the quality of bathing water (76/160/EEC), *Official J. E. C.*, 5 February 1976, L31/1.

CEC. 1992. 'Towards Sustainability - A European Programme of Policy and Action in Relation to the Environment and Sustainable Development'. CON(92)23, European Commission, Brussels.

CEC,1994. 'Packaging and Packaging Waste Directive', DIR(94/62/EC), European Commission, Brussels.

CEC. 1997. 'Cost-Benefit Analysis of the Different Municipal Solid Waste Management Systems : Objectives and Instruments for the Year 2000', European Commission, Brussels.

Centre for Energy Conservation and Environmental Technology. 1998. A European Environmental Aviation Charge: Feasibility Study, Delft.

CEPN. 1992. *CEC/US project on the external costs of fuel cycles - nuclear accidents* Centre D'Etude Sur L'Evaluation de la Protection dans le Domaine Nucleaire.

Chestnut, L., Colome, L., Keller, L., Lambert, W., Ostro, B., Rowe R. and Wojciechowski, S. 1988, *Heart Disease Patients' Averting Behavior, Costs of Illness, and Willingness to Pay to Avoid Angina Episodes,* Report to Office of Policy Analysis, US Environmental Protection Agency, Washington DC.

Chestnut, L. and Patterson, A. *Human Health Benefits Assessment of the Acid Rain Provisions of the 1990 Clean Air Act Amendments*, August 1994, for US EPA.

Cline. W. 1993. The Economics of Climate Change, Institute for International Economics, Washington DC.

Cochran, R. and Cotton, A.W. 1985. *Municipal water demand study*, Oklahoma City and Tulsa, Oklahoma, Water Resources, Research, Volume 21, No 7, July.

COHERENCE, NTUA (Athens) and Centre for Economic Studies (Leuven) .1997. *Economic Evaluation of Community Options to Limit CO*<sub>2</sub> *Emissions at the Horizon 2005 and 2010*, Final Report, European Communities, Luxembourg.

Coker, A., Tunstall, S., Penning-Rowsell, E. 1989. *An Evaluation of the recreation and amenity benefits of a flood alleviation scheme for Maidenhead*, FHRC, Middlesex University, England.

Collins. A., Evans. A. 1994. Aircraft Noise and Residential Property Values: an Artificial Neural Network Approach, *Journal of Transport Economics and Policy*, May, 175-197.

Colman. D. 1989. Economic Issues from the Broads Grazing Marshes Conservation Scheme, *Journal* of Agricultural Economics, 40(3).

Convery. F., Rooney. S. 1996. Making Markets Work for the Economy and the Environment – lessons from experience in Greece, Ireland, Portugal and Spain, in *Environmental Taxes and Charges – National Experiences and Plans*, papers from the Dublin Workshop, European Foundation for the Improvement of Living and Working Conditions, Luxembourg.

Coopers Lybrand. CSERGE. EFTEC. 1996. Cost-benefit Analysis of the Different Municipal Solid Waste Management Systems, DGXI, European Commission, Brussels.

Coopers Lybrand. 1997. Landfill Tax: is it Working?, HMSO, London.

Crooks. S., Ledoux. L. 1999. *Mitigation Banking as a Tool for Strategic Coastal Zone Management: a UK Perspective*, CSERGE, University of East Anglia, Paper GEC 99-02.

Cropper, M. and Sussman, F. 1988. Families and the Economics of Risk to Life, *American Economic Review*, 255-260.

Cropper, M. and Sussman, F. 1990. Valuing Future Risks to Life, *Journal of Environmental Economics and Management*, 19, 1990, 160-174.

Cropper, M. and Freeman, A. 1991. Environmental Health Effects, in J.Braden and C.Kolstad (eds), *Measuring the Demand for Environmental Quality*, North Holland, New York.

Cropper, M., Aydede, S. and Portney, P. 1994. Preferences for Life Saving Programs: How the Public Discounts Time and Age, *Journal of Risk and Uncertainty*, 8, 243-265

Cropper, M. and Simon, N.B. 1994. Are We Valuing Risks to Life Correctly ? World Bank, Washington DC, *mimeo*.

Crosson. P., Stout. A. 1983. *Productivity Effects of Cropland Erosion in the United States*, Resources for the Future, Washington DC.

Crosson. P. 1995. *Soil Erosion and Its On-Farm Productivity Consequences: What Do We Know*?, Discussion Paper 95-29, Resources for the Future, Washington DC.

CSERGE. 1993. *Externalities from Landfill and Incineration*, Centre for social and economic research for the Global Environment (CSERGE), University College London with Warren Spring Laboratory and Economics for the Environment Consultancy (EFTEC), HMSO, London.

CSERGE and EFTEC Ltd. 1994. A Survey of Noise Cost Estimates, CSERGE, mimeo.

CSERGE and EFTEC Ltd. 1998. *Setting Environmental Priorities in the European Union*, Earthscan, London, forthcoming.

CSERGE. 1998. *Global Environment, Working Paper GEC 96-17, CSERGE, University College London and University of East Anglia.* 

CSERGE. 1998. The Clean Development Mechanism: Benefits of the CDM for Developing Countries, *Executive Summary*, Centre for Social and Economic Research on the Global Environment, CSERGE, London.

Danielson, L.E. 1979. An analysis of residential demand for water using micro time-series data, Water Resources Research, volume 15, No 4, August.

De Lansink. O., Peerlings. J. 1995. From Specific Impacts of Policy Changes on pesticide use in Dutch Arable Farming, *Papers from the Wageningen Workshop on Pesticides*, Wageningen Agricultural University, August 1995.

Delanche. X. 1996. Implementing Ecotaxes in France: Some Issues, in *Environmental Taxes and Charges* – *National Experiences and Plans*, papers from the Dublin Workshop, European Foundation for the Improvement of Living and Working Conditions, Luxembourg.

Department of Economics. University of Oslo.

Desai. N., Mathur. S. 1996. *Market-Based Instruments for the Implementation of the Montreal Protocol in Developing Countries*, Environment Department Paper 053, World Bank, Washington DC.

Desaigues, B. and Rabl, A. 1995. Reference Values for Human Life: An Econometric Analysis of a Contingent Valuation in France, in N Schwab Christie and N.Soguel (eds), *Contingent Valuation, Transport Safety and the Value of Life*, Kluwer, Boston and Dordrecht, 85-112.

Desvousges, W. et al., 1993. Review of Health Effects Resulting from Exposure to Air Pollution, Research Triangle Institute, Task Force on Externality Costing, Working Paper 1 (revised), November.

DETR (UK Department of the Environment, Transport and Regions). 1996. *Economic Instruments for Water Pollution*, together with Welsh Office, Department of the Environment for Northern Ireland and the Scottish Office, London.

Diffey. B. 1992. Stratospheric ozone depletion and the risk of non melanoma skin cancer in a British population, Phys Med Biol, 37, 2267-2279.

Dixon. R., Andrasko. K., Sussman. F., Trexler. M., Vinson. T. 1993. Forest Sector Carbon Offset Project: Near-Term Opportunities to Mitigate Greenhouse Gas Emissions, *Water, Air and Soil Pollution*.

Dockery, D.W., Schwartz, J. and Spengler, J. 1992. Air Pollution and Daily Mortality: Association with Particulates and Acid Aerosols, *Environmental Research*, 59, 362-373.

Dockery, D.W. Pope III, C.A., Xiping, X. et al., 1993. An Association Between Air Pollution and Mortality in Six U.S. Cities, *The New England Journal of Medicine*, 329(4), pp 1753-1808.

Douglas. W. 1995. Market-based Instruments in Germany and the Netherlands: a Case of Competition vs. Co-operation? *European Environment*, 5,1-6.

Downing, M., and Ozuna, T. (1996). Testing the reliability of the benefit function transfer approach, *Journal of Environmental Economics and Management*, 30, 316-322.

Drake. L. 1991. The Non-Market Value of the Swedish Agricultural Landscape, *European Review of Agricultural Economics*, 19, 351-364.

DRI. 1994. Potential Benefits of Intergration of Environmental and Economic Policies, Graham and Trotman for the CED, London.

DRI. 1997. The Energy, Environment and Economic Effects of Phasing Out Coal Subsidies in OECD Countries, OECD, Paris.

Dubgaard. A. 1987. Anvendelse af afgiter til regulering af pesticidforbruget, (Taxation as a means to control pesticide use), Report 35, Statens Jordgrugsøkonomiske Institut, Copenhagen.

Dubourg. R. 1995. *Material Damages Caused by Air Pollution: a Note*, Centre for Social and Economic Research on the Global Environment, University College London, *mimeo*.

Dunford. R. 1992. Natural resource damages from oil spills. Tietenberg. T. H (ed). Innovation in environmental policy and economic and legal aspects of recent developments in environmental enforcement and liability. New Horizons in Environmental Economics Series. 165-93.

EBRD,1999. *The Nuclear Safety Account: Central and Eastern Europe and the former Soviet Union,* EBRD, London.

ECNC (European Centre for Nature Conservation). 1998. Facts and Figures on Europe's Biodiversity, ECNC, Tilburg.

Ecofys. 1998. *Reduction in the Emission of HFCs, PFCs and SF6 in the European Union*, Ecofys, Utrecht, Netherlands.

ECOTEC. 1993a. An Evaluation of the Benefits of Reduced Sulphur Dioxide Emissions: A Contingent Valuation Study of Upland Vegetation, report to the UK Department of the Environment - Annex 4, Birmingham.

ECOTEC. 1993b. An Evaluation of the Benefits of Reduced Sulphur Dioxide Emissions: A Contingent Valuation Study of Aquatic Ecosystems, report to the UK Department of the Environment-Annex 4, Birmingham.

ECOTEC. 1997. *Economic Instruments for Pesticide Minimisation*, Report to the UK Department of the Environment, London.

ECOTEC. 1997. UK Landfill Tax Study, Part II, Phase A, ECOTEC, Birmingham.

EEA. 1998. Europe's Environment: the Second Assessment, Elsevier, Oxford.

EEA. 1999. Europe's Environment: the Second Assessment, EEA, Copenhagen.

Eeckhoudt. L., Schieber. C., Schneider. T. 1997. *Expected Utility Approach: Risk Aversion and Collective Risk*, CEPN, ExternE.

EFTEC and CSERGE. 1999. Valuing Preferences for Changes in Water Abstraction from the River Ouse, Yorkshire Water, restricted, Economics for the Environment Consultancy, London.

EFTEC. 1996. *Research into Damage Valuation Estimates for Nitrogen Based Pollutants, Heavy Metals and Persistent Organic Compounds*, Department of the Environment, London. 3 volumes, Economics for the Environment Consultancy, London.

EFTEC. 1999. Review of Technical Guidance on Environmental Appraisal, Department of Environment, Transport and Regions, UK, Economics for the Environment Consultancy.

Ekins. P. 1996. How Large A Carbon Tax is Justified by the Secondary Benefits of CO2 Abatement ? *Resource and Energy Economics*, **18**, 161-187.

Ekins. P., Speck. S. 1998. Competitiveness and Exemptions from Environmental Taxes in Europe,

Environmental and Resource Economics, 13: 369 - 396, 1999.

Elvingson. P. 1998. System of differentiated dues (for shipping in Swedish ports), Acid News, March. *Energy Policy*, 24 (7) 609-619.

ERM and University of Newcastle. 1997. Economic Appraisal of the Environmental Costs and Benefits of Potential Solutions to Alleviate Low Flows in Rivers, Report to S W Region of Environment Agency.

Espey. M. 1998. Gasoline Demand Revisited: an International Meta-analysis of Elasticities, *Energy Economics*, 20.

European Commission, 1996. Cost Benefit Analysis of the Different Municipal Solid Waste Management Systems: Objectives and Instruments for the Year 2000, European Commission, Brussels.

European Commission. 1995. Europeans and the Environment, a survey conducted in the context of the Eurobarometer 43.1 bis INRA (Europe) ECO.

European Commission. 1995. A Welfare Cost Assessment of Various Measures to Reduce Pollutant Emissions from Passenger Road Vehicles for the Year 2010, report prepared by Touche Ross.

European Commission. 1996. *Tax Provisions with a Potential Impact on Environmental Protection*, Luxembourg.

European Commission. 1997. *Voluntary Approaches*, Environmental Policy Research Briefs, Number 1, EC.

European Commission. 1998. Communication on Climate Change - Towards an EU Post Kyoto Strategy, COM(98)353.

European Commission. 1998. *ExternE*: Externalities of Energy, Office for Official Publications of the European Commission, Luxembourg.

European Commission, 1999, Towards a European Intergrated Coastal Zone Management (ICZM) Strategy: General Principles and Policy Options, and Lessons from the European Commissions's Demonstration Programme on ICZM 1997-1999, European Commission, Brussels.

European Environment Agency. 1995. Europe's Environment: The Dobris Assessment, Earthscan Publications, London.

European Environment Agency. 1996. Environmental Taxes, Implementation and Environmental Effectiveness, Environmental Issues series no.1, Copenhagen.

European Environmental Agency. 1997. Environmental Agreements: Environmental Effectiveness, EEA, Copenhagen.

European Environment Agency, 1998. Methodology Report, *Baseline Projections of Selected Waste Streams*, prepared by the European Topic Centre on Waste.

Eurostat. 1990. Arable land areas in EU15 countries 1990. Eurostat.

Eurostat. 1995. Europe's Environment: Statistical Compendium for the Dobris Assessment, Eurostat, Luxembourg.

Eurostat, 1997, Demographic Indicators. Eurostat.

Evans. R. 1996. Soil Erosion and Its Impacts in England and Wales, Friends of the Earth, London.

Ewers. H-J., Rennings. K. 1995. Economics of nuclear risks - a German study, in O. Hohmeyer. and R.

ExternE. 1995. Externalities of Energy: Vol.2: Methodology, DGXII, European Commission, Brussels.

ExternE. 1999. *Maintenance, Improvement, Extension and Application of the ExternE Accounting Framework*, ExternE, <u>http://ExternE.jrc.es/</u>

Eyre. N., Downing. T., Hoekstra. R., Rennings. K., Tol. R., 1997. *Global Warming Damages*, Final Report of the ExternE Global Warming Sub-Task, DGXII, European Commission, Brussels.

Faeth. P., Westra. J. 1993. Alternatives to Corn and Soybean Production in Two Regions of the United States, in P.Faeth (ed), *Agricultural Policy and Sustainability*, World Resources Institute, Washington DC.

Fankhauser. S. 1995. *Valuing Climate Change: the Economics of the Greenhouse*, Earthscan, London. Fankhauser. S., Tol. R., Pearce. D.W. 1996. Equity and the Aggregation of the Damage Costs of Climate Change in V.Nacicenovic, W Nordhaus, R Richels and F Toth (eds), *Climate Change: Integrating Science*,

Fankhauser. S., Tol. R., Pearce. D.W. 1997. The Aggregation of Climate Change Damages: a Welfare Theoretic Approach, *Environment and Resource Economics*, Vol.10, No.3, October, 249-266. *Economics and Policy*, International Institute for Applied Systems Analysis, Laxenburg, Austria, 167-178.

Fankhauser. S., Tol. R., Pearce. D.W. 1997. Extensions and Alternatives to Climate Change Impact Valuation: on the Critique of IPCC Working Group III's Impact Estimates, *Environment and Development Economics*, forthcoming.

Feder. B. J. 1996. Lower bids seen at sale of rights to pollute air, New York Times 145, (50,375) pp. 20,22.

Fernandez .V., Soufi. R., Tuddenham. M. 1995. The Tax on Undeveloped Land in France, in R. Gale and S Barg (eds), *Green Budget Reform*, Earthscan, London.

Fischoff, B., Slovic, P. and Lichtenstein, S. 1979, Weighing the Risks, *Environment*, 21(4), 17-20 and 32-38.

Fisher, A., Chestnut, L. and Violette.D. 1989. The Value of Reducing Risks of Death: a Note on New Evidence, *Journal of Policy Analysis and Management*, 8(1), 88-100.

Flanders, D., Haddix, A., Olson, D., Romieu, I., White, M., Williamson. G.D. 1995. *Review of Urban Air Pollution Health Impact Methodology*, Center for Disease Control, Atlanta.

Flores, N. and Carson, R. 1997. The Relationship Between the Income Elasticities of Demand and Willingness to Pay, *Journal of Environmental Economics and Management*, 33, 287-295

Foster, H.S. and Beattie, B.R. 1979. Urban residential demand for water in the United States, Land Economics, volume 55, No 1, February.

Foster. V., Mourato. S., Ozdemorglu. E., Pearce. D. W. 1998. Incorporating external impacts in pest management choices, in W Vorley and D Keeney (eds), *Bugs in the System: Redesigning the Pesticide Industry for Sustainable Agriculture*, Earthscan, London, 94-106.

Franzen, A C,1996. *The Pesticide Reduction Programme in Sweden: Update*, Worldwide Fund for nature, Gland.

Fredman. P. 1994, Values of an Endangered Species- The Case of the White-backed Woodpacked Woodpacker in Sweden, Sveriges Lantbruksuniversitet, Institutionen for Skogsekonomi, no 181.

Freeman III, A.M. 1986. On Assessing the State of the Art of the Contingent Valuation Method of Valuing Environmental Changes, in R.G.Cummings, D.Brookshire and W.Schulze (eds), *Valuing Environmental Goods: an Assessment of the Contingent Valuation Method*, Rowman and Allenheld, Totowa, NJ. 1986, 148-161.

Freeman III, A.M.1993. *The Measurement of Environmental and Resource Values*, Resources for the Future, Washington DC.

Frohburg. K. 1995. Programs for stimulating extensive agriculture in Germany, in M. Galy. H. 1998. *Nuclear Risks in Eastern Europe*, M.Sc Dissertation, Economics Department, University College London.

Garrod. G.D., Willis. K.G. 1991. The Environmental Economic Impact of Woodland: a Two Stage Hedonic Price Model of the Amenity Value of Forestry in Britain, ESRC Countryside Change Initiative, working paper 19.

Garrod. G.D., Willis. K.G. 1991. The Hedonic Price Method and the Valuation of Countryside Characteristics, ESRC Countryside Change Initiative, working paper 14. Garrod., G.D., Willis. K.G. 1991. Some Empirical Estimates of Forest Amenity Value, ESRC Countryside Change Initiative, working paper 13.

Garrod. G. D., Willis. K. 1994. An Economic Estimate of the Effect of a Waterside Location on Property Values, *Environmental and Resource Economics*, 4, 209-217.

Garrod. G.D., Willis. K. 1994. Valuing Biodiversity and Nature Conservation at a Local Level, *Biodiversity and Conservation*, 3, 555-565.

Garrod GD, KG Willis, 1995, Valuing the Benefits of the South Downs Environmentally Sensitive Area, *Journal of Agricultural Economics*, 46(2) 160-173.

Garrod. G., Willis. K. 1996. Estimating the benefits of environmental enhancement: a case study of the Rover Darent, *Journal of Environmental Planning and Management*, 39, 2, 189-203.

Garrod. G., Willis. K. 1997. *Estimating Lost Amenity Due to Landfill Waste Disposal*, Journal Resource, Conservation and Recycling 22 (1998) 83-95.

Garrod. G.D., Willis. K.G. 1997. The Non-use Benefits of Enhancing Forest Biodiversity: a Contingent Ranking Study, *Ecological Economics*, 21, 45-61.

Georgiou, S., Langford, I., Bateman. I., Turner. R.K. 1996. 'Determinants of individuals' WTP for reductions in environmental health risks: a case study of bathing water quality'. CSERGE Working Paper 96-14, UEA.

GEP Environnement. 1997. CREDOC and Centre de Prospective et de Veille Scientifique, 1997. *Recherche et Environnement: Thèmes Prioritaires et Thèmes Emergents*, Report for the European Commission DGXI, *restricted*.

Gibbons. D. 1986. *The Economic Value of Water*, Resources for the Future, Washington DC. Gielen. D., Koutstaal. P., Kram. T., Rooijen Van S. 1998. *Post Kyoto: Effects on the Climate Policy of the European Union*, Netherlands energy Research Foundation (ECN), ECNC-98-040.

Glomsrød. S., Godal. O., Henriksen. J., Haagenrud. S., Skancke. T. 1996. *Air Pollution: Impacts and Values: Corrosion Costs of Building Materials and Cars in Norway*, Report 96-03, Norwegian Pollution Control Authority, Oslo.

Goulder. L.H., Parry. I.W.H., Williams. R.C.III., Burtraw. D. 1995. *The Cost Effectiveness of Alternative Instruments for Environmental Protection in a Second Best Setting*, Resources for the Future, Washington DC, USA.

Graham, J., Chang, B-H. and Evans, J. 1994. Poorer is Riskier. *Risk Analysis*, **12**, 3, 333-337 Graham. B., Hinchy. M., Fisher. B., Tulpul—. N. 1998. Climate change negotiations – the Kyoto Protocol, *Outlook 98*, ABARE, Canberra.

Green. C.H., Tunstall. S.M. 1990b. The recreational and amenity value of river corridors, paper given at *Conservation and Management of Rivers-An international Conference*. York.

Gregory, R.1986. Interpreting Measures of Economic Loss: Evidence from Contingent Valuation and Experimental Studies, *Journal of Environmental Economics and Management*, Vol.13, No.4, 325-337.

Gregory. K., Mathews. A., Newton. A., Nind. A. 1992. The potential impact of a 10\$ barrel energy / carbon tax on UK carbon dioxide emissions, Climate Change: Designing a Practical Tax System, OECD, Paris.

Gren, A-M. Folke, C., Turner, K. and Bateman, I. 1994. Primary and Secondary Values of Wetland Ecosystems, *Environmental and Resource Economics*, 4, No.1, February, 55-74.

Gren. I.M., Folke. C., Turner. K., Bateman. I. 1994. Primary and Secondary Values of Wetland Ecosystems, *Environmental and Resource Economics* 4:55-74.

Gren. I.M., Groth. K.H., Sylven. M. 1995. Economic Values of Danube Floodplains, *Journal of Environmental Management*, 45, 333-345.

Griffin, R.C. and Chang, C. 1985. *Pretest analyses of water demand in thirty communities*, Water Resources Research, Volume 26, No 10, October.

Grubb. M., Vrolijk. C. 1997. *Defining and trading emission commitments in the Kyoto agreement*, Royal Institute of International Affairs, September.

Grue. B., Langeland. J., Larsen. O. 1997. *Housing Prices: Impacts of Exposure to Road Traffic and Location*, TOI Report, 351, Oslo.

Gusbin. D., Klaasen. G., Kouvaritakis. N. (forthcoming). Costs of a Ceiling on Kyoto Flexibility, *Energy Policy*.

Halcrow and Partners .1996. *Water Resource Planning: Strategic Options*, Report to UK National Rivers Authority, Bristol.

Hammitt. J. 1997. Stratospheric Ozone Depletion, in R.Morgenstern (ed), *Economic Analyses at EPA:* Assessing Regulatory Impact, Resources for the Future, Washington DC, 131-170.

Hammitt. J., Thompson. K. 1997. Protecting the Ozone Layer, in J.Graham and J.Hartwell (eds). *The Greening of Industry: a Risk Management Approach*, Harvard University Press, Cambridge, Mass, 43-92.

Hanemann, M.1991, Willingness to Pay and Willingness to Accept: How Much Can They Differ ?, *American Economic Review*, 81, 635-647.

Hanke, S. and de Mare, L. 1982. *Residential water demand. A pooled time series, cross section study of Malmo, Sweden, Water Resources Bulleting,* Volume 18, No 4, August.

Hanley. N. 1989. Problems in Valuing Environmental Improvements Resulting from Agricultural Policy Changes: the Case of Nitrate Pollution, in Dubgaad A & Nielsen A (eds) *Economic Aspects of Environmental Regulations in Agriculture*, Wissenschaftsverlag, Vauk Kiel, Kiel.

Hanley. N., Craig. S. 1991. Wilderness development decisions and the Krutilla-Fisher model: The case of Scotlands flow country. *Ecological Economics*. (4). 145-164.

Hanley. N.D., Ruffell. R. 1992. The valuation of forest characteristics. Discussion Paper 92/10. Department of Economics. University of Stirling.

Hanley. N., Spash. C., Walker. L. 1995. Problems in Valuing the Benefits of Biodiversity Protection, *Environmental and Resource Economics*, 5, 249-272.

Hanley. N., Kirkpatrick. H., Simpson. I., Oglethorpe. D., MacDonald. A. 1996. Ecological-Economic Modelling of the Conservation of Threatened habitats: heather Moorland in the Northern Isles of Scotland, *Biodiversity and Conservation* 5, 1207-1219.

Hanley. N., MacMillan. D., Wright. R.E., Bullock. C., Simpson. I., Parsisson. D., Crabtree. R. 1997. Contingent Valuation versus Choice Experiments: Estimating the Benefits of Environmentally Sensitive Areas in Scotland, mimeo.

Haywood. S., Robinson. C., Heady. C. 1991. COCO-1: Model for assessing the cost of offsite consequences of accidental releases of radioactivity. NRPB-R243, National Radiological Protection Board, Harwell, UK.

Heintz. R., Tol. R. 1996. Secondary benefits of climate control policies: implications for the Global Environment Facility, *Centre for Social and Economic Research on the Environment (CSERGE), University College London.* 

Hellberg. J. 1994. The Swedish experience of a charge, on emissions of nitrogen oxides from energy production, where the revenue from the charge is refunded to the group of plants that pay it, Swedish Environmental Protection Agency (SEPA).

Henderson. A.. 1996. Waste Charges and Taxes in the Netherlands, in European Foundation for the Improvement of Living and Working Conditions, *Environmental Taxes and Charges: National Experiences and Plans*, EFILWC, Dublin.

Henry. D., Olson. A. 1992. *A Review of Commonwealth Grants to Voluntary Conservation Organisations*, prepared for the Australian Commonwealth Minister for Arts, Sport, the Environment, Tourism and Territories.

Herrington. P. 1997. The Proper Pricing of Water, in T O'Riordan (ed), *Ecotaxation*, Earthscan, London. HM Customs and Excise. 1998. *Review of the Landfill Tax: Report*, HM Customs and Excise, Newcastle.

Hoen, H. and Winther, G. 1991. Attitudes to and w.t.p for multiple use forestry and preservation of coniferous forests in Norway (mimeo), Agricultural University of Norway.

Hoerner. J. A. 1995. Tax Tools for Protecting the Atmosphere: The US Ozone-Depleting Chemicals Tax, in R Gale, S Barg and A Gillies (eds), *Green Budget Reform*, Earthscan, London, 185-199.

Hoerner. J. A. 1996. Taxing Pollution, in E Cook (ed), *Ozone Protection in the United States*, World resources Institute, Washington DC, 39-54.

Hofreither and S Vogel (eds), *The Role of Agricultural Externalities in High Income Countries*, Wissenschaftsverlag Vauk Kiel KG, 161-180.

Hoglund. L. 1997. Estimation of Household Demand for Water in Sweden and Its Implications for Potential Tax on Water Use, *Unit for Environmental Economics, Department of Economics, Gothenburg University, Paper 1997:12.* 

Holland M. and Krewit, W.1996. *Benefits of an Acidification Strategy for the European Union*, European Commission, DGXI, Brussels.

Holland. M., Krewitt. W. 1997. *Benefits of an acidification strategy for the European Union*, European Commission DGX11 Joule programme.

Hornung. M. 1996. In *Green Accounting in Europe: the Role of Damage Estimation: Four Case Studies*, final report of a project under the JOULE2 Programme of DGXII of the European Commission.

Hotte M H, van der Vlies J., Hafkamp W. A. 1995b. Levy on Surface Water Pollution in the Netherlands, in Gale R and S Barg with A Gillies (eds) *Green Budget Reform: An International Casebook of Leading Practices*, Earthscan, London.

Hotte. M. H., van der Vlies. J., Hafkamp. W. A.1995a. Dutch Policies Aimed at Diminishing Mineral Surpluses from Manure and Fertilizer, in Gale R and S Barg with A Gillies (eds) *Green Budget Reform: An International Casebook of Leading Practices*, Earthscan, London.

Hubert. P., Barny. M., Moatti. J. 1991. Elicitation of decision makers' preferences for management of major hazards, *Risk Analysis*, Vol 11, No. 2.

ICONA. 1991. *Plan National de Lutte Contre L'Erosion*, Institute National Pour la Conservation de la Nature, Madrid [this document has not been traced].

IIASA, AEA Technology, DNMI and RIVM. 1998. Economic Evaluation of Air Quality Targets for Tropospheric Ozone, DGXI, Brussels.

Intergovernmental Panel on Climate Change. 1992. Climate Change 1992: The Supplementary Report of the IPCC Scientific Assessment, Cambridge University Press, Cambridge.

Intergovernmental Panel on Climate Change (IPCC). 1996. Working Group 3, *Economic and Social Dimensions of Climate Change*, Cambridge University Press, Cambridge.

Intergovernmental Panel on Climate Change. 1996. *Climate Change 1995: The Science of Climate Change*, Cambridge University Press, Cambridge.

IPCC. 1994. Radiative forcing of Climate change: the 1994 Report of the Scientific Assessment Working Group of IPCC, UNEP, WMO.

ISSP,1993. *Response to the International Social Survey Program 1993*. Zentralarchiv fuer Empirische Sozialforschung, Koeln, Germany.

IVM, NILU and IIASA, 1997. *Economic Evaluation of Air Quality for Sulphur Dioxide, Nitrogen Dioxide, Fine and Suspended Particulate Matter and Lead*, Report to DGXI, European Commission, Brussels.

James. A., Green. M.1998. Indicators of Investments in Biological Diversity Conservation: National Expenditure on Protected Areas, World Conservation Monitoring Centre, Cambridge.

Johannesson, M., and P-O Johansson, 1995a. Is the Value of a Life Year Gained Independent of Age?, Stockholm School of Economics, Stockholm, *mimeo*.

Johannesson, M., and P-O Johansson, 1995b. Quality of Life and the WTP for an Increased Life Expectancy at an Advanced Age, *Working Papers in Economics and Finance, No.85*, Stockholm School of Economics.

Johannesson. M., Johansson. P.O. 1996. To Be or Not To Be, That Is the Question: An Empirical Study of the WTP for an Increased Life Expectancy at an Advanced Age, *Journal of Risk and Uncertainty*, 13, pp 163-174.

Johansson and Zavisic, 1989. Sweden, preserve all natural forest/virgin forest in Sweden, in Svenska folkets mijobudget, Ekomisk Debath, 6, 472 - 474.

Johansson. P. O., Kristom. B. 1988. Study cited in MacMillan. D. (1996) valuing the environmental benefits of reduced acid deposition in the semi-natural environment. PhD dissertation. University of Stirling.

Johansson. P. O., Kristrom. B., Maler. K. G., 1989. Welfare evaluations in contingent valuation experiments with discrete response data: Comment. *American Journal of Agricultural Economics*, 71 (4), 1054-56.

Johansson, P-O. 1995, *Evaluating Health Risks: an Economic Approach*, Cambridge University Press, Cambridge.

Johansson. P.O. 1996. The external costs of road transport in Sweden, in D.Maddison *et al*, 1996. Jones. H., Howson. G.U., Rosengren-Brink., Hornung. M., Linareas. P. 1997. *Review of the Effects of Air Pollutants on Agricultural Crops, Forestry and Natural Vegetation*, Institute of Terrestrial Ecology, Report T707074f5, ITE, Merlewood.

Jones-Lee, M., M.Hammerton and P Philips, 1985. The Value of Safety: Results of a national Sample Survey, *The Economic Journal*, 95, 49-72.

Jones-Lee, M.1989. The Economics of Safety and Physical Risk, Blackwell, Oxford.

Jones-Lee, M.1991. Altruism and the Value of Other People's Safety, *Journal of Risk and Uncertainty*, 4:2, 213-219

Jones-Lee, M.1992. Paternalistic Altruism and the Value of Statistical Life, *The Economic Journal*, 102:410, 80-90.

Jones-Lee, M., G.Loomes, D.O'Reilly, P.Philips, 1993. *The Value of Preventing Non-fatal Road Injuries: Findings of a Willingness to Pay National Sample Survey*, Working Paper WP/SRC/2, Transport Research Laboratory, Crowthorn.

Jones-Lee, M., G.Loomes, P.Philips, 1995. Valuing the Prevention of Non-fatal Road Injuries: Contingent Valuation vs Standard Gambles, *Oxford Economic Papers*, 47.

Kahn. J. 1996. Economic Instruments to Abate Acidification in Sweden, in European foundation for the Improvement of Living and Working Conditions, 1996, *Environmental taxes and charges: National Experiences and Plans*, EFILWC, Dublin.

Kahneman, D. and A.Tversky, 1979, Prospect Theory: an Analysis of Decisions Under Risk, *Econometrica*, 43, 263-291

Katsouyanni. K. 1996. The APHEA Project. Short Term Effects of Air Pollution on Health: a European Approach using Epidemiological Time Series Data, *Journal of Epidemiology & Community Health*, vol 50 Supplement 1, whole supplement.

Katsouyanni. K., Touloumi.G., Spix. C., Schwartz. J., Balducci.F., Medina. S., Rossi. G., Wojtyniak. B., Sunyer. J., Bacharova. L., Schouten. J. P., Ponka. A., Anderson. H. R. 1997. Short Term Effects of Ambient Sulphur Dioxide and Particulate matter on Mortality in 12 European Cities: Results from Time Series Data from the APHEA Project, *British Medical Journal*, 314, 1658-63.

Keeney, R. 1990. Mortality Risks Induced by Economic Expenditures, Risk Analysis, 10, 1, 147-159

Keeney, R. 1994. Mortality Risks Induced by the Costs of Regulation, *Journal of Risk and Uncertainty*, 8, 95-110

Keeney, R. 1997. Estimating Fatalities Induced by the Economic Costs of Regulation, *Journal of Risk and Uncertainty*, 14, 5-23

Kellert, S. 1996. The value of life: Biological Diversity and Human Society, Island Press, Washington, DC, USA.

Kidholm, K. 1995. Assessing the Value of Traffic Safety Using the Contingent Valuation Technique: the Danish Survey, in N Schwab Christie and N.Soguel (eds), *Contingent Valuation, Transport Safety and the Value of Life*, Kluwer, Boston and Dordrecht, 45-61

King. D., Wainger. L. 1999. Assessing the Economic Value of Biodiversity Using Indicators of Site Conditions and Landscape Context, Paper to OECD Experts Meeting on the Economic value of Biodiversity, OECD, Paris, October 1999.

Kirchhoff, S., Colby, B., and LaFrance, J. (1997). Evaluating the performance of benefits transfer: an empirical inquiry, *Journal of Environmental Economics and Management*, 33, 75-93

Klaasen. G., Forsund. F. (eds). 1994. Economic Instruments for Air Pollution Control, Kluwer Academic Publishers, Dordrecht.

Klaasen. G. 1996. Acid Rain and Environmental Degradation: the Economics of Emission Trading, Edward Elgar, Cheltenham England.

Klaasen. G., Nentjes. A. 1997. Creating markets for air pollution control in Europe and the USA, Environmental and Resource Economics, 19: 125-146.

Kopke. A. 1996. An Economic Approach to EU Water Policy, ENVECO Meeting, Brussels.

Kopp, R.1993. Environmental Economics: Not Dead But Thriving, *Resources*, Spring, No.111, 7-12. Kosz. M. 1996. Valuing Riverside Wetlands: the Case of the "Donau-Auen" National Park, *Ecological Economics* 16, 109-127.

Koutstaal. P., Nentjes. A. 1995. Tradable Carbon Permits in Europe: Feasibility and Comparison with Taxes, Journal of Common Market Studies, Vol 33, No.2, 1995.

Kraemer. A. R. 1995. Water Resources Taxes in Germany, in Gale R and S Barg with A Gillies (eds) *Green Budget Reform: An International Casebook of Leading Practices*, Earthscan, London.

Kristr\_m, B. and Riera, P. 1996, Is the Income Elasticity of Environmental Improvements Less Than One ? *Environmental and Resource Economics*, **7**, 45-55

Kristrom. B. 1990. A non-parametric approach to the estimation of welfare measures in discrete response valuation studies. *Land Economics*, 66, (2), 135-39.

Krupnick, A. 1986. *A Preliminary Benefits Analysis of the Control of Photochemical Oxidants*, US Environmental Protection Agency, Washington DC.

Krupnick, A., and M.Cropper, 1989. *Valuing Chronic Morbidity Damages: Medical Costs, Labour Market Effects and Individual Valuations*, Report to Office of Policy Analysis, US Environmental Protection Agency, Washington DC.

Krupnick, A. and M.Cropper, 1992, The Effect of Information on Health Risk Valuations, *Journal of Risk and Uncertainty*, 5, 29-48.

Krupnick, A., K.Harrison, E.Nickell and M.Toman, 1993. *The Benefits of Ambient Air Quality Improvements in Central and Eastern Europe: a Preliminary Assessment*, Discussion Paper ENR93-19, Resources for the Future, Washington DC

Krupnick, A., Harrison, K., Nickell, E., and Toman, M. (1996). The value of health benefits from ambient air quality improvements in Eastern Europe: an exercise in benefits transfer, *Environmental and Resource Economics*, 7, 307-332.

Kuitunen. M., Tormala. T. 1994. Willingness of Students to Favour the Protection of Endangered Species in a Trade-off Conflict in Finland, *Journal of Environmental Management*, 42, 111-118. L¬vgren. K. 1994. Economic Instruments for Air Pollution Control in Sweden in Klaasen G and Forsund F R (eds), *Economic Instruments for Air Pollution Control*, Kluwer Academic Publishers, Dordrecht.

Lambert, J. 1992. Impact des transportes terrestres sur l'environnement, methodes d'euduation et couts sociaux, Synthese Interets, 23, Paris.

Laroui. F., Leeuwen. Van M. 1996. Fiscal policy and greenhouse gases: the case of the Netherlands, *Ecological Economics*, 18.

Larsen. B., Shah. A. 1992. *World Fossil Fuel Subsidies and global Carbon Emissions*, The World Bank, Washington DC.

Laukkanen, R. 1981. *Flow forecasts in general planning of municipal water and sewage works*, Water Research Institute Publication No.41, National Board of Water, Helsinki.

Lawrence. D. G. 1998. *Aspects of Soil Protection as Seen by the Commission*, Bonn Workshop on Soil Protection Policies within the European Union, 9-11 December.

Le Goffe. P. 1995. 'The benefits of improvements in coastal water quality: a contingent approach', *Journal of Environmental Management*, vol 45, pp305-317.

Leder. A. 1996. Taxes with an Environmental Base and the Green Commission, in *Environmental Taxes* and *Charges – National Experiences and Plans*, papers from the Dublin Workshop, European Foundation for the Improvement of Living and Working Conditions, Luxembourg.

Lee. D. 1996. Trading Pollution, in E Cook (ed), *Ozone Protection in the United States*, World resources Institute, Washington DC, 31-38.

Leek. F., de. Savorning. A. 1996. Charges in Water Quality and Management in the EU, *European Environment*, Vol 6, 33-39.

Lehoczki. Z. 1999. Co-ordinating environmental and fiscal policy in Hungary: possibilities and constraints, in K Schlegelmilch (ed), *Green Budget Reform in Europe: Countries at the Forefront,* Springer-Verlag, Berlin, 149-163.

Linddal M and A. Navrud. S. 1996. *Willingness to Pay for Health Symptoms, Increased Fish Populations due to Reduced Acidification and Road Traffic Noise: Results from a Contingent Valuation Study,* report to the Norwegian Pollution Control Authority, ENCO Environmental Consultants (in Norwegian).

Loehman, E.T., S.Berg, A.Arroyo, R.Hedinger, J.Schwartz, M.Shaw, R.Fahien, V.Dhe, R.Fishe, D.Rio, W.Rossley and A.Green, 1979. Distributional Analysis of Regional Benefits and Costs of Air Quality Control, *Journal of Environmental Economics and Management*, **6**, 222-243. London Economics. 1990. *The Effects of Taxes on Fertilisers*, consultancy report, London.

Longstreth. J. F., de Grujii. M., Kripke. M., Takizawa. Y., Van der Leun. J. 1995. Effect of Increased Solar Radiation on Human Health, *Ambio*, Vol.XXIv, No.3, May, 153-165.

Loomis, J. (1992). The evolution of a more rigorous approach to benefit transfer: benefit function transfer, *Water Resources Research*, 28, 3, 701-705.

Loomis. J.B., White. D.S. 1996. Economic Benefits of Rare and Endangered Species: summary and Meta-analysis, *Ecological Economics*, 18, 197-206.

Lutter, R. and J.Morrall, 1994. Health-Health Analysis: A New Way to Evaluate Health and Safety Regulation, *Journal of Risk and Uncertainty*, 8, 43-66.

Machado, F. and Mourato, S. 1999. 'Improving the Assessment of Water Related Health Impacts: Evidence from Coastal Waters in Portugal.' CSERGE Working Paper GEC 99-09. MacMillan. D., Hanley. N., Buckland. S. 1995. Valuing Biodiversity Losses due to Acid Deposition: a Contingent Valuation Study of Uncertain Environmental Gains, Discussion paper in Ecological Economics, no 95/01, Dept Economics, Univ. Stirling.

Maddison. D. 1994. The Shadow Price of Greenhouse Gases and Aerosols, CSERGE, University College London, *mimeo*.

Maddison, D.O. Johansson, T.Littman, E.Verhoef and D.W.Pearce, 1996. *Blueprint 5: The Social Costs of Transport*, Earthscan, London.

Maddison. D., Pearce. D. W., Johansson. O., Calthrop. E., Litman. T., Verhoef. E. 1996. *Blueprint 5: The True Costs of Road Transport*, Earthscan, London.

Maddison. D. 1997. The Economic Value of Visibility: a Survey, CSERGE, mimeo.

Maddison. D. 1997. Valuing Changes in Life Expectancy in England and Wales Caused by Ambient Concentrations of particulate Matter, CSERGE, University College London.

Maddison. D., Pearce. D.W., Lvovsky. K. 1997. *Damage Costs from Fuel Use in Major Urban Conurbations*, Environment Department, World Bank, Washington DC (restricted). Maddison, D. 1998. Valuing changes in life expectancy in England and Wales caused by ambient concentrations of particulate matter, Centre for Social and Economic Research on the Global Environment (CSERGE), University College London, *mimeo*.

MAFF Economics and Statistics Group. 1999. *Reducing farm subsidies – Economic adjustment in rural areas*, Working Paper 2, MAFF: London.

Maier, G., S.Gerking and P.Weiss, 1989. The Economics of Traffic Accidents on Austrian Roads: Risk Lovers or Policy Deficit ? *Mimeo*, Wirtschaftuniversitat, Vienna, cited in Rowlatt et al, 1998.

Malska. P., Luukkanan. J., Vehmas. J., Kaivo-oja. J. 1997. *Environment based Energy Taxation in the Nordic countries*, Finnish Ministry of the Environment, Helsinki.

March Consulting. 1999. UK Emissions of HFCs, PFCs and SF6 and Potential Emission Reduction Options, Department of the Environment, Transport and Regions, London.

Markandya. A. 1997. *Monetary Valuation Issues in Extended ExternE*, for EC DGX11 (JOULE programme), European Commission, Brussels.

Mauch. S., Rothengatter.W. 1994. *External Effects of Transport*, Final Report to International Union of railways (UIC), Paris.

McMichael, A., R.Anderson, P.Elliott, P.Wilkinson, A.Ponce de Leon and F Simon Soria (the 'London Review Group'), 1995. *Review of Methods Proposed, and Used, for Estimating the Population Health Risks of Exposure to Urban Air Pollution*, London School of Hygiene and Tropical Medicine, London.

Mendelsohn. R., Morrison. W., Schlesinger. M., Andronova. N. 1996. *Global Impact Model for Climate Change*, Unpublished mss, School of Forestry, Yale University.

Mendelsohn. R., Neumann. J. (eds). 1997. *The Impact of Climate Change on the US Economy*, Cambridge University Press, Cambridge.

Miller, T. 1989. Willingness to Pay Comes of Age: Will the System Survive ?, *Northwestern University Law Review*, 83, 876-907.

Miller, T. and J.Guria, 1991. *The Value of Statistical Life in New Zealand*, New Zealand Ministry of Transport, Wellington NZ.

Mitchell. R., Carson. R. 1986. Valuing Drinking Water Risk Reductions Using the Contingent Valuation

Method, Report to the US EPA, Washington DC.

Moret, Ernst and Young (1996) *Tax Provisions with a Potential Impact on Environmental Protection*, Report to DGXI, Brussels.

Morris. J., Phillips. P., Read. A. 1998. The UK landfill tax: an analysis of its contribution to sustainable waste management, *Resources, Recycling and Conservation*, **23**, 259-270.

Moor De A.P.G., 1997. *Perverse Incentives*, Institute for Research on Public Expenditure, Report to the Earth Council, San Jose.

Moor De A.P.G, Calamai. P. 1997. Subsidising Unsustainable Development, Undermining the Earth with Public Funds, Earth Council Report, San Jose.

Moor De A.P.G., Van Beers. C. 1998. Scanning Subsidies and Policy Trends in Europe and Central Asia, working paper, Institute for Research on Public Expenditure, Netherlands.

Mourato. S. 1997. *Effects of Air Pollution on Ecosystems: A Survey of Economic Valuation Studies*, report prepared for the UNECE Task Force on Economic Aspects of Abatement Strategies, Madrid, December.
Murdoch. J., Thayer. M. 1990. The benefits of reducing the incidence of non melanoma skin cancers: a defensive expenditures approach, *Journal of Environmental Economics and Management*, **18**, 107-119. Naskali (Eds), proceedings of the Workshop valuing Biodiversity. Espoo, Finland, October 1992, Scandinavian Forest Economics 34.

Navrud. S. 1989. Estimating Social Benefits of Environmental Improvements from Reduced Acid Depositions: A Contingent Valuation Survey, in Folder H and van Ireland E (eds), *Valuation Methods and Policy making in Environmental Economics*, Ellesmere Science Publishers, Amsterdam.

Navrud. S. 1992. (Ed.), Pricing the European Environment, Scandinavian University Press.

Navrud. S. 1997. Valuing Health Impacts from Air Pollution in Europe. New Empirical Evidence on Morbidity, Department of Economics, Agricultural University of Norway, mimeo.

Needleman, L.1976, Valuing Other People's Lives, *The Manchester School*, 44, 309-342. Nentjes. A., Koustaal. P., Klaassen. G. 1995. *Tradable Carbon Permits: Feasibility, Experiences, Bottlenecks*, University of Groningen, Report No 410 100 114.

NERA and CASPAR, 1997. *Valuation of Deaths from Air Pollution*, Report to the Department of the Environment, Transport and Regions.

Nieswiadomy. M. L., Molina. D. J. 1989. Comparing residential water demand estimates under decreasing and increasing block rates using household data. *Land Economics*. 65, (3). 280-89.

Nordhaus. W. 1991. To Slow or Not to Slow: The Economics of the Greenhouse Effect. *Economic Journal*, 101, 407, 920-937.

Nordhaus. W. 1994. *Managing the Global Commons: the Economics of Climate Change*, MIT Press, Cambridge, Mass.

Nordic Council of Ministers. 1996. The Use of Economic Instruments in Nordic Environmental Policy, Nordic Council, Copenhagen.

O'Doherty, R. 1994. Contingent Valuation as a Participatory Process, Department of Economics, University of West England, Bristol, *mimeo*.

O'Grady Sharrock. G. 1995. *The Nitrate Issue: prospects for taxing the use of nitrogen fertiliser in the UK*, dissertation, MSc in Environmental and Resource Economics, University College London, London.

OECD. 1980. Pollution Charges in Practice, OECD, Paris.

OECD. 1986. Fighting Noise, Strengthening Noise Abatement Policies, Paris.

OECD. 1991. Environmental Policy: How to Apply Economic Instruments, Paris.

OECD. 1991. Fighting Noise in the 1990s, OECD, Paris.

OECD. 1992. Economic accounts for agriculture 1977 - 1990, OECD, Paris.

OECD. 1994, The Economics of Climate Change: Proceedings of an OECD/IEA Conference, OECD, Paris.

OECD. 1994a. Managing the Environment: the role of economic instruments, Paris.

OECD. 1994b. Environment and Taxation: the cases of the Netherlands, Sweden and the United States, Paris.

OECD. 1995. Environmental Taxes in OECD Countries, Paris.

OECD. 1995. Global Warming: Economic Dimensions and Policy Responses, OECD, Paris.

OECD. 1996. Saving Biological Diversity: Economic Incentives, OECD, Paris.

OECD. 1997. Evaluating Economic Instruments for Environmental Policy, OECD, Paris.

OECD. 1997. Reforming Energy and Transport Subsidies: Environmental and Economic Implications, OECD, Paris.

OECD. 1998. Reforming Energy and Transport Subsidies, OECD, Paris.

OECD. European Conference of Ministers of Transport (ECMT). 1998. *Efficient Transport for Europe: Policies for Internalisation of External Costs*, OECD, Paris.

Oldeman. L., Hakkeling. R., Sombroek. W. 1991. World Map of the Status of Human Induced Soil Degradation, ISRIC, UNEP, The Hague.

Olderman. R. 1992. *Global Extent of Soil Degradation*, in Biennial Report of the International Soil Reference and Information Centre Report 1991-1992, Wageningen, Netherlands.

Olivecrona C, 1996, The nitrogen oxide charge on energy production in Sweden, in Gale R and Barg S (eds), *Green Budget Reform*, Earthscan, London.

Olsthoorn. X. et al., 1997. Economic Evaluation of Air Quality for Sulphur Dioxide, Nitrogen Dioxide, Fine and Suspended Particulate Matter and Lead, Report to DGXI, Brussels, August.

Oskam, A. J. 1995. The Economics of Pesticides: an Overview of the Issues, *Papers from the Wageningen Workshop on Pesticides*, Wageningen Agricultural University, August 1995.

Ostro, B.1994. *Estimating Health Effects of Air Pollution: a Method with an Application to Jakarta*, Policy Research Department, Working Paper 1301, World Bank, Washington DC.

Ostro, B. 1995b. Air Pollution and Mortality: Results from Santiago, Chile, Policy Research Department, Working Paper 1453, World Bank, Washington DC.

Ottinger. (eds), Social Costs of Energy: Present Status and Future Trends, Springer-Verlag, Berlin, 150-166.

Otway, H. and J.Cohen, 1975, *Revealed Preferences: Comments on the Starr Benefit-Risk Relationships*, Research Memorandum 75-5, International Institute for Applied Systems Analysis, Laxenburg, Vienna.

Parsons, G., and Kealy, M. (1994). Benefits transfer in a random utility model of recreation, *Water Resources Research*, 30, 8, 2477-2484

Pearce, D.W. 1980. The Social Incidence of Environmental Costs and Benefits, in T.O'Riordan and R.K.Turner (eds), *Progress in Resource management and Environmental Planning*, Vol.2, John Wiley and Son, London.

Pearce, D.W.1986. Cost Benefit Analysis, (2nd edition), Macmillan, Basingstoke.

Pearce. D. W., Auditing the Earth, Environment, 40 (2), March, 1988, 23-28.

Pearce, D.W. A.Markandya and E.Barbier, 1989. Blueprint for a Green Economy, Earthscan, London.

Pearce. D.W. 1992. *The Secondary Benefits of Greenhouse Gas Control*, CSERGE Working Paper 92-12, CSERGE, University College London.

Pearce, D.W. and R.K.Turner, 1992. The Ethical Foundations of Sustainable Economic Development, *Advances in Human Ecology*, 1, 177-195.

Pearce, D.W., C.Bann and S.Georgiou, 1992. The Social Cost of Fuel Cycles, HMSO, London.

Pearce, D.W. and I.Brisson, 1993. BATNEEC: the Economics of Technology-Based Environmental Standards, *Oxford Review of Economic Policy*, Vol.9, No.4, Winter, 24-40.

Pearce. D.W., Bann. C. 1993. *The Social Costs of Fuel Cycles*, HMSO, London. Pearce. D. W., Moran. D. 1994. *The Economic Value of Biodiversity*, Earthscan, London.

Pearce, D.W., D.Whittington and S.Georgiou, 1994. *Project and Policy Appraisal: Integrating Economics and Environment*, Organisation for Economic Cooperation and Development, Paris.

Pearce. D. W., Moran. D. 1994. The Economic Value of Biodiversity, Earthscan, London.

Pearce. D.W. 1996. Economists and Climate Change, Environment and Planning, Vol.29, 1-4.

Pearce. D.W., Crowards. T. 1996. Particulate Matter and Human Health in the United Kingdom,

Pearce. D.W., Cline. W.R., Achanta. A., Fankhauser. S., Pachauri. R., Tol. R., Vellinga. P. 1996. The Social Costs of Climate Change: Greenhouse Damage and the Benefits of Control, in Intergovernmental Panel on Climate Change, *Climate Change 1995: Economic and Social Dimensions of Climate Change*, Cambridge University Press, Cambridge, 183-224.

Pearce. D.W., Tinch. R. 1998. The true price of pesticides, in W Vorley and D Keeney (eds), *Bugs in the System: Redesigning the Pesticide Industry for Sustainable Agriculture*, Earthscan, London, 50-93.

Pearce. D.W. 1998. Valuing Statistical Lives: Briefing Paper for: Economic Assessment of Priorities for a European Environmental Policy Plan, CSERGE, University College London and EFTEC, London.

Pearce. B., Pearce. D.W. 1999. *Setting Environmental Taxes for Aircraft: A Case Study of the UK*, Seminar presentation to the Economic Regulation Group Civil Aviation Authority, CSERGE, University College London.

Pearce. D. W. 1999. Water Pricing: Conceptual and Theoretical Issues, paper for European Commission and Instituto da Agua, Portuguese Ministry of Environment Conference: *Pricing Water: Economics, Environment and Society*, CSERGE, University College London.

Peck. S., Teisberg. T. 1993. Global Warming Uncertainties and the Value of Information: An Analysis Using CETA, *Resource and Energy Economics*, **15**, 1, 71-97.

Penning-Rowsell. E., Green. C., Thompson. P., Coker. A., Tunstall. S., Richards. C., Parker. D. 1992. 'The economics of coastal management', Belhaven Press, London.

Persson, U., and M.Cedervall, 1991. The Value of Risk Reduction: Results of a Swedish Sample Survey, *IHE Working Paper 1991:6*, Swedish Institute of Health Economics.

Persson, U., A Ligner Norinder and M.Svensson, 1995. Valuing the benefits of Reducing the Risk of Non-Fatal Road Injuries: the Swedish Experience, in N Schwab Christie and N.Soguel (eds), *Contingent Valuation, Transport Safety and the Value of Life*, Kluwer, Boston and Dordrecht, 63-83

Pezzey. J., Mill. G. 1998. *A Review of Tariffs for Public Water Supply*, Environment Department, University of York, York, UK.

Pimentel, D., C.Wilson, C.McCullum, R.Huang, P.Dwen, J.Flack, Q.Tran, T.Saltman, B.Cliff, 1996, *Environmental and Economic Benefits of Biodiversity*, College of Agriculture and Life Sciences, Cornell University, Ithaca, NY, *mimeo*.

Pimentel. D., Acguay. H., Biltonen. M., Rice. P., Silva. M., Nelson. J., Lipner. V., Giordano. S., Harowitz. A., d'Amore. M. 1992. Environmental and economic costs of pesticide use, *Bioscience*, **42**, 10, 750-760.

Pocock. R.L., Thomas. R.M. 1995. *Review of the Recycling Credits Scheme*, in Journal of Waste Management and Resource Recovery, Volume 2, Number 1, 1995.

Pommerehne. K. 1988. Measuring Environmental Benefits: Comparison of a Hedonic technique and CVM, in D.B\_s, M.Rose and C.Seidl, (eds), *Welfare and Efficiency in Public Economics*, Berlin.
Pope III, C.A. and D.Dockery, 1992. Acute Health Effects of PM<sub>10</sub> Pollution on Symptomatic and Asymptomatic Children, *American Review of Respiratory Disease*, Vol.145, 1123-1128.
Pope III, C.A. J.Schwartz, and M.Ransom, 1992. Daily Mortality and PM<sub>10</sub> Pollution in Utah Valley, *Archives of Environmental Health*, Vol.47, No.3, 211-217.

Pope III, C.A., M.J.Thun, M.Namboordi, D.W.Dockery, J.D.Evans, F.Speizer and C.W.Heath Jr, 1995. Particulate Air Pollution as a Predictor of Mortality in a Prospective Study of US Adults, *American Journal of Respiratory and Critical Care Medicine*, Vol.151, 669-674.

Portney, P. and R Stavins, 1994. Regulatory Review of Environmental Policy: the Potential Role of Health-Health Analysis, *Journal of Risk and Uncertainty*, 8, 111-122.

Press. J., S-derqvist. T. 1998. On estimating the benefits of groundwater protection: a contingent valuation study in Milan, in *Regulating chemical accumulation in the environment: The integration of toxicology and economics in environmental policy making*, Swanson T and Vighi M (eds) Cambridge University Press.

Pruckner. G. 1995. Agricultural Landscape Cultivation in Austria: an Application of the CVM, *European Review of Agricultural Economics*, 22, 173-190. Quantifying the Effects of Fine Scale Exposure, Dose and Effects of Ozone: Part 2 - Estimating Yield Losses for Agricultural Crops, *Water, Air and Soil Pollution*.

Rabl. A. 1996. 'Discounting of Long-Term Costs: What Would Future Generations Prefer Us to Do? *Ecological-Economics*; 17, 3, 137-45.

Rabl. A., Eyre. N. 1997. *An Estimate of Regional and Global O3 Damage from Precursor NOx and VOC Emissions*, Ecole des Mines, Paris and Eyre Energy and Environment, Carnforth.

Rabl. A., Spadaro. J., McGavran. P. 1998. Health Risks of Air Pollution from Incinerators: a Perspective, *Waste Management Research*, forthcoming.

Rico R, 1995, The US Allowance Trading System for Sulphur Dioxide, *Environmental and Resource Economics*, 5.

RIVM. 1995. Dutch priority programme on Acidification: Ammonia, the Facts, RIVM, Bilthoven.

RIVM. 1998. Integrated Environmental Assessment of the Base Line Scenario for the EU State of the Environment 1998 Report, RIVM, Bilthoven.

Rosegrant. M., Gazmuri. R.1994. *Reforming Water Allocation Policy Through Markets in Tradable Water Rights: Lessons from Chile, Mexico and California*, International Food Policy Institute, Washington DC.

Rowe, R., L.Chestnut and W Shaw, 1984, Oxidants and Asthmatics in Los Angeles: a Benefits Analysis, in S Duk Lee (ed), *Evaluation of the Ozone/Oxidants Standard*, Air Pollution Control Association, Pittsburgh.

Rowe, R., L.Chestnut, D.Peterson and C.Miller, 1986, *The Benefits of Air Pollution Control in California, Vols 1 and 2*, Energy and resource Consultants, Boulder for California Air resources Board.

Rowe, R. and L.Chestnut, 1986, *Oxidants and Asthmatics in Los Angeles: a Benefits Analysis*, Report to Office of Policy Analysis, US EPA, EPA-230-09-86-018, Washington DC.

Rowe, R. and T.Nethercut, 1987, *Economic Assessment of the Impact of Cataracts*, RCH Hagler Bailly, Colorado, *mimeo*.

Rowe. R., Lang. C., Chestnut. L., Latimer. D., Rae. D., Bernow. S., White. D. 1994. New York State Environmental Externalities Cost Study, Vol.1, Oceana Press, New York.

Rowe. R., L.Chestnut., C.Lang., S.Bernow and D.White, 1995. *The New York Environmental Externalities Cost Study: Summary of Approach and Results*, Paper presented to European Commission, International Energy Agency and Organisation for Economic Cooperation and Development Workshop on External Costs of Energy, Brussels, January 30-31, 1995.

Rowlatt, P., M.Spackman, S.Jones, M.Jones-Lee and G.Loomes, *Valuation of Deaths from Air Pollution*, NERA and CASPAR, Report to UK Department of the Environment, Transport and Regions, February. Saderqvist. T. 1994. The costs of meeting a drinking water quality standard: the case of Atrazine in Italy, in L.Bergman and D.Pugh (eds), *Environmental Toxicology: Economics and Institutions*, Kluwer, Dordrecht, 151-171.

Saderqvist. T. 1995. *The Benefits of Reduced Eutrophication of the Baltic Sea: a Contingent Valuation Study*, Stockholm School of Economics, Stockholm, *mimeo*.

Saelensminde. K., Hammer. F. 1994. Assessing Environmental Benefits by Means of Conjoint Analysis, Institute of Transport Economics, Oslo.

Saelensminde. K. 1997. Contingent Valuation of Urban Traffic Air Pollution and Noise, Institute of Transport Economics, Oslo.

Sagoff, M. 1993. Environmental Economics: an Epitaph, *Resources*, Spring, No.111, 2-7. Salmons. R. 1998. *Economic Instruments and the Business Use of Energy*, Centre for Social and Economic Research on the Global Environment (CSERGE), University College London.

Sändstrom . M. 1994. *A Discrete Choice Model of Swedish Beach Recreation*, Stockholm School of Economics, Stockholm, *Mimeo*.

Sch←rer. B. 1994. Economic Instruments in Air Pollution Control: The Case of Germany, Federal Environment Agency, Germany.

Schelling. T. 1999. Intergenerational discounting in Portney. P and Weyant. J (eds) *Discounting and Intergenerational Equity*, Resources for the Future. Washington DC 99-102.

Scheraga. J., Leary. N. 1994. Costs and Benefits of Using Energy Taxes to Mitigate Global Climate Change, in *Proceedings of the Annual Conference of the National Tax Association*, Washington DC.

Schnutenhaus. J. 1996. Tax differentials for catalytic converters and unleaded petrol in Germany, in Gale R and Barg S (eds), *Green Budget Reform,* Earthscan, London.

Schwab Christe. N. and N Soguel, 1995, *The Pain of Victims and the Bereavement of their Relatives: a Contingent Valuation Experiment*, IDHEAP, University of Lausanne, Switzerland, *mimeo*.

Schwartz, J. and A Marcus, 1990. Mortality and Air Pollution in London: a Time Series Analysis, *American Journal of Epidemiology*, Vol. 131, No.1, 1990, 185-194

Schwartz, J., C.Spix, H.Wichman and E.Malin, 1991. Air Pollution and Acute Respiratory Illness in Five German Communities, *Environmental Research*, Vol.56, 1-14.

Schwartz, J.1991a. Particulate Air Pollution and Daily Mortality in Detroit, *Environmental Research*, Vol.56, 204-213.

Schwartz, J. 1991b. Particulate Air Pollution and Daily Mortality: a Synthesis, *Public Health Reviews*, Vol.19, 39-60.

Schwartz, J. and D.Dockery, 1992a. Particulate Air Pollution and Daily Mortality in Steubenville, Ohio, *American Journal of Epidemiology*, Vol.135, 12-19.

Schwartz, J. and D.Dockery, 1992b. Increased Mortality in Philadelphia Associated with Daily Air Pollution Concentrations, *American Review of Respiratory Disease*, Vol.145, 1992, 600-604

Schwartz, J. 1993a. Air Pollution and Daily Mortality in Birmingham, Alabama, *American Journal of Epidemiology*, Vol.137, 1136-1147.

Schwartz, J. 1993b. Particulate Air Pollution and Chronic Respiratory Disease, *Environmental Research*, Vol.62, 7-13.

Schwartz, J. D.Slater, T.Larson, W.Pierson and J.Koenig, 1993. Particulate Air Pollution and Hospital Emergency Room Visits for Asthma in Seattle, *American Review of Respiratory Disease*, Vol.147, 826-831.

Schwartz, J.1994. Air Pollution and Daily Mortality: A Review and Meta Analysis, *Environmental Research*, 64, pp 36-52.

Sea Empress Environmental Evaluation Committee. 1998. *The Environmental Impact of the Sea Empress Oil Spill*, The Stationery Office, London.

Seaton, A., W.Macnee, K.Donaldson and D.Godden [1995], Particulate Air Pollution and Acute Health Effects, *The Lancet*, Vol.345, January 21, 176-178.

Shlegelmilch. 1998. Energy Taxation in the EU and Some Member States: Looking for Opportunities Ahead, Wuppertal Institute.

Simpson. D. 1993. Photochemical model calculations over Europe for two extended summer periods: 1985 and 1989: model results and comparisons with observations, *Atmospheric Environment*, 27A, 921-943.

Skou Anderson. M. 1997. Assessing the effectiveness of Denmark's waste tax, *Environment*, 40, 4, 11-15 and 38-41.

Slaper. H., Blaauboer. R. 1997. *Risk assessment for accidental releases from nuclear power plants in Europe*, National Institute of Public Health and the Environment, Bilthoven, The Netherlands. Small, K. and C.Kazimi, 1995. On the Cost of Air Pollution from Motor Vehicles, *Journal of Transport Economics and Policy*, January, 7-32.

Smith. S., Pearson. M. 1991. *The European Carbon Tax, An Assessment of The European Commission's Proposals*, The Institute for Fiscal Studies (IFS), London. Smith. V.K. 1992. Environmental costing for agriculture: will it be standard fare in the Farm Bill of 2000 ? *American Journal of Agricultural Economics*,

Soguel. N. 1994. Évaluation Monétaire des Atteintes à l'Environnement: une Étude Hedoniste et Contingente sur l'Impact des Transports, University of Neuchâtel, Switzerland.

Sorrell. S. 1994a. Pollution on the market: the US experience with emissions trading for the control of air pollution, Science Policy Research Unit, University of Sussex, Brighton, England.

Sorrell. S. 1998. Why Sulphur Trading failed in the UK, Science and Policy Research Unit (SPRU), electronic working paper series, Paper no 19.

Spaninks. F. A. 1993. Een schatting van de sociale baten van beheersovereenkomsten met behulp van de contingent valuation method'. *MSc thesis*. Department of Agricultural Economics and Policy, Waginen Agricultural University.

Spash. C., Hanley. N. 1995. Preferences, Information and Biodiversity Preservation, *Ecological Economics*, 12, 191-208.

Stanners. D., Bourdeau. P. (eds). 1995. Europe's Environment, The Dobris Assessment, prepared by the European Environment Agency Task Force for the European Commission: DGXI and Phare.

Starr, C. 1972. Benefit Cost Studies in Sociotechnical Systems, in Committee on Public Engineering Policy, *Perspective on Benefit-Cost Decision-Making*, National Academy of Sciences, Washington DC.

Steiner. R., McLaughlin. L., Faeth. P., Janke. R. 1995. Incorporating externality costs into productivity measures: a case study using US agriculture, in V Barbett, R Payne and R Steiner (eds), *Agricultural Sustainability: Environmental and Statistical Considerations*, Wiley, New York, 209-230

Sterner. T., Bartelings. H. 1996. Household Waste Management in a Swedish Municipality: Determinants of Waste Disposal, Recycling and Composting.

Strand. J. (1981). Valuation of fresh fish populations as a public good in Norway. Working paper. Svendsen. G.T. 1997. A general model for  $CO_2$  regulation: the case of Denmark, Energy Policy, Vol. 26.

Svendsen. G.T. 1998. Towards a  $CO_2$  market in the EU: the case of electric utilities, *European Environment*, **8**, 121-128.

Swedish Environmental Protection Agency (SEPA). 1992. Economic instruments for sulphur and nitrogen oxides control in Sweden, SEPA, Stockholm.

Swedish Environmental Protection Agency (SEPA). 1993. Five Economic Instruments in Swedish Environmental Policy, SEPA, Stockholm.

Swedish Environmental Protection Agency (SEPA). 1998. *Environmental Taxes in Sweden*, Report 4745, SEPA, Stockholm.

Szereny. Z. 1997. The Application of Environmental Valuation Methods in Hungary: the case of Bukk National Park, Discussion paper in ecological economics, no 97/05, Dept. Economics University of Stirling.

Teir. G. 1996. Evolution of  $CO_2$  / energy taxes in Finland, in European foundation for the Improvement of Living and Working Conditions, 1996, *Environmental Taxes and Charges: National Experiences and Plans*, EFILWC, Dublin.

TemaNord . 1994. *The Use of Economic Instruments in Nordic Environmental Policy*, report No: 561, TemaNord, Copenhagen.

TFEEAS, 1999. *Review of expreience in applying economic instruments to the abatement of nitrogen oxides, sulphur dioxide, volatile organic compounds and ammonia,* report for the 15<sup>th</sup> meeting of the Task Force on Economic Aspects of Abatement Strategies, Rome, Italy June 1999, for the Convention on Long-Range Transboundary Air Pollution.

Thackray. J. 1996. An overview of current water use and charges in the UK compared to other countries, in Chartered Institution of Water and Environmental Managers, *Water Conservation and Re-Use*, CIWEM, London.

Tinch. R. 1995. The Valuation of Externalities, Report to the UK Department of Transport, London.

Tolley, G.S., L.Babcock, M.Berger, A.Bilotti, G.Blomquist, R.Fabian, G.Fishelson, C.Kahn, A.Kelly, D.Kenkel, R.Kumm, T.Miller, R.Ohsfeldt, S.Rosen, W.Webb, W.Wilson and M.Zelder, 1986. *Valuation of Reductions in Human Health Symptoms and Risks*, University of Chicago report to US EPA, Washington DC.

Tuddenham. M. 1995. The System of Water Charges in France, in Gale R and S Barg with A Gillies (eds) *Green Budget Reform: An International Casebook of Leading Practices*, Earthscan, London.

Turner, R.K. 1993. Sustainability: Principles and Practice, in R.K.Turner (ed), *Sustainable Environmental Economics and Management*, Belhaven, London, 3-36.

Turner. R. K., Ozdemiroglu. E., Steele. P. 1995. Environmentally Sensitive Areas in the UK: Economic Incentives for Sustainable Farming, in R. Gale and S Barg (eds), *Green Budget Reform*, Earthscan, London.

Turner. R. K. 1995. *The Baltic Drainage Basin Report*, EV5V-CT-92-0183, European Commission, Brussels.

Turner. R. K. 1997. *Managing Nutrient Fluxes and Pollution in the Baltic: an Interdisciplinary Simulation Study*, Centre for Social and Economic Research on the Global Environment, University of East Anglia and University College London, GEC Paper 97-17.

UK Department of Environment, 1993. Statistical Bulletin, Survey of Public Attitudes to the Environment, UK 1993, Department of Environment, London.

UK House of Lords, Select Committee on the European Communities, Session. 1994, 1995. 1<sup>st</sup> Report. *Bathing Water*, III Paper 6-1, and Session 1994 1995, 7<sup>th</sup> Report, *Bathing Water Revisited*, HL Paper 6-1, London: HMSO.

UNFAO, 1999. UNFAOSTAT Database Online.

US EPA. 1987. *Regulatory Impact Analysis: Protection of Stratospheric Ozone*, Office of Air and Radiation, US EPA, Washington DC.

USEPA, 1995, Acid Deposition Standard Feasibility Study Report to Congress, EPA 430/R-95-001a, October 1995, Washington, USA.

USEPA, 1999. *The benefits and Costs of the Clean Air Act 1970 - 1990*, US Environment Protection Agency (EPA), Washington, DC, USA.

Utell, S. and J.Samet, 1993. Particulate Air Pollution and Health: New Evidence of an Old Problem, *American Review of Respiratory Disease*, Vol.147, 1334-35. Vainio. M. 1995. *Traffic Noise and Air Pollution: Valuation of Externalities with Hedonic Price and Contingent Valuation Methods*, Helsinki School of Economics and Business Administration, Doctoral Dissertation.

Van den Bergh. J., Button. K., Nijkamp. P., Pepping. G. 1997. Effectiveness of pesticide price policies in agriculture, Ch 12 of J van den Bergh, K Button, P Nijkamp and G Pepping *Meta-Analysis in Environmental Economics*, Kluwer, Dordrecht, 163-174.

Vatn, A. and D.Bromley, 1994. Choices Without Prices Without Apologies. *Journal of Environmental Economics and Management*, 26, 129-148.

Verhoef. E. 1996. The external costs of road transport in the Netherlands, in D.Maddison *et al.*, 1996. *Blueprint 5: The True Costs of Road Transport*, Earthscan, London.

Viscusi, W.K., W.Magat and A.Forrest, 1988. Altruistic and Private Valuations of Risk Reduction, *Journal of Policy Analysis and Management*, Vol.7, No.2, 227-245.

Viscusi, W.K., W.Magat, and J.Huber, 1991, Pricing Environmental Health Risks: Survey Assessments of Risk-Risk and Risk-Dollar Trade Offs for Chronic Bronchitis, *Journal of Environmental Economics and management*, 21 (1), 32-51.

Viscusi, W.K. 1992. Fatal Trade-Offs: Public and Private Responsibilities for Risk, Oxford University Press, Oxford and New York

Viscusi, W.K., and R.Zeckhauser, 1994. The Fatality and Injury Costs of Expenditures, *Journal of Risk and Uncertainty*, 8, 19-41.

Viscusi, W.K.1994. Risk-Risk Analysis. Journal of Risk and Uncertainty, 8, 5-17.

Viscusi. W.K. 1995. The automobile risk metric for valuing health risks, in N Schwab Christe and N Soguel (eds), *Contingent Valuation, Transport Safety and the Value of Life*, Kluwer, Dordrecht, 171-193.

Vliet. Van A., Heunks. C., Brink. ten B. 1999. Natural Areas in Europe and the Change of Some Pressures, RIVM, Bilthoven.

Völker. T. 1998. The Economics of Soil Erosion: an Assessment of the On-Site and Off-Site Costs of Soil Erosion for England and Wales, M. Sc Thesis, University College London.

Weinberger. M., Thomassen. H., Willeke. R. 1991. *Kosten des Lärms in der Bundesrepublik Deutschland*, Umweltbundesamt, Berichte 9/91, Berlin.

Welsh. H. 1996. Recycling of carbon / energy taxes and the labour market, *Environmental and Resource Economics*, 8.

Wexler. P. 1995. *Iowa's 1987* Groundwater Protection Act, in Gale R and S Barg with A Gillies (eds) *Green Budget Reform: An International Casebook of Leading Practices*, Earthscan, London.

White. P.R., Franke. M., Hindle. P. 1995. Integrated Solid Waste Management, A Lifecycle Inventory, Blackie Academic & Professional.

Willis. K., Benson. J. (1989). Recreational values of forests. Forestry, 62. 93-110.

Willis. K., Garrod. G. 1991. Landscape Values: a Contingent Valuation Approach and Case Study of the Yorkshire Dales National Park, working paper 21, ESRC Countryside Change Initiative.

Willis. K., Garrod. G.D., Sheperd. P. 1996. *Towards a methodology for costing biodiversity conservation in the UK*, A Report to the Department of the Environment, London HMSO.

Willis. K.G. 1990. Valuing Non-market Wildlife Commodities: an Evaluation and Comparison of Benefits and Costs, *Applied Economics*, 22, 13-30.

Willis. K.G. 1991. The Recreational Value of the Forestry Commission Estate in Great Britain: a Clawson-Knetsch Travel Cost Analysis, forthcoming in *Scottish Journal of Political Economy*.

Willis. K.G., Benson. J.F. 1988. A Comparison of User Benefits and Costs of Nature Conservation at Three Nature Reserves, *Regional Studies*, vol 22.5, 417-28.

Willis. K.G., Benson. J.F. 1988. Valuation of Wildlife: a Case Study on the Upper Teesdale Site of Special Scientific Interest and Comparison of Methods in Environmental Economics, in K.Turner, ed., *Sustainable Environmental Management*, Belhaven Press, London.

Willis. K.G., Garrod. G., Dobbs. I. 1990. The Value of Canals as a Public Good: the Case of the Montgomery and Lancaster Canals, ESRC Countryside Change Initiative, working paper 5.

Willis. K.G., Garrod. G., Saunders. C.M. 1993. Valuation of the South Downs and Somerset Levels and Moors Environmentally Sensitive Area Landscapes by the General Public. A report to the Ministry of Agriculture, Fisheries and Food. Centre for Rural Economy, University of Newcastle upon Tyne.

Willis. K.G., Garrod. G.D., Benson. J.F., Carter. M. 1996. Benefits and Costs of the Wildlife Enhancement Scheme: a Case Study of the Pevensey Levels, *Journal of Environmental Planning and Management*, 39 (3), 387-401

Wilson. K. 1992. *The Diversity of Life*, Allen Lane, The Penguin Press, London. Wohlgemuth. N. 1997. *World Transport Energy Demand Modelling: Methodology and elasticities*, Energy Policy, vol 25, Nos 14-15, pp 1109-1119.

Working Paper - Energy Series Paper No. 41, World Bank, Washington, DC.

World Bank. 1991. Health and safety aspects of nuclear power plants, Industry and Energy Dept.

World Bank. 1995. *Monitoring Environmental Progress, A Report on World in Progress*, The World Bank, Washington DC.

World Resources Institute, 1994. World Resources 1994-1995, Oxford University Press, Oxford and New York.

Xu, X., J.Gao, D.Dockery, and Y.Chen, 1994. Air Pollution and Daily Mortality in Residential Areas of Beijing, China, *Archives of Environmental Health*, Vol.49, No.4, July/August, 216-222 Young. R. 1996. *Measuring Economic Benefits for Water Investments and Policies*, Technical Paper 338, World Bank, Washington DC.

Zorpette. G. 1994. 'A slow start for emissions trading, IEEE Spectrum July, 49-52.

Zylicz. T., Bateman. I., Georgiou. S., Markowska. A., Dziegielewska. D., Turner. R. K., Graham A., Langford . I. 1995a. Contingent Valuation of Eutrophication Damage in the Baltic Sea Region, CSERGE Working Paper 95-03, CSERGE, University of East Anglia.

Zylicz. T., Bateman. I., Georgiou. S., Markowska. A., Dziegielewska. D., Turner. R. K., Graham A., Langford . I. 1995a. Contingent Valuation of Eutrophication Damage in the Baltic Sea Region, CSERGE Working Paper 95-03, CSERGE, University of East Anglia.